Periodic and open classical spin Calogero-Moser chains

Nicolai Reshetikhin

ABSTRACT. We construct classical Hamiltonian system similar to spin Calogero-Moser type systems. They can be regarded as a many particle system with composite spin degrees of freedom and also as an integrable spin chain of Gaudin type. We prove that these Hamiltonian systems are superintegrable.

Introduction

1. Classical Calogero-Moser (CM) systems were among the first integrable N-particle systems of one dimensional particles [3][25] with the potential $1/(q_i-q_j)^2$. This model was generalized to the potential $1/sh^2(q_i-q_j)$ in [36]. Then it was extended to other root systems and to elliptic potentials in [28], to a model involving spin degrees of freedom in [16].

There is an extensive literature on spin versions of CM systems. For example in [21][22] solutions to equations of motion to the elliptic spin Calogero-Moser system were related to special elliptic solutions to the matrix KP hierarchy. The relation to gauge theories were explored in many papers, see for example [27][15]. A variety of spin CM systems were obtained by L. Feher, see for example [10][11][12], in particular he derived important examples related to homogeneous spaces. Two spin CM systems were studied in [18][19]. Integrable chains of relativistic spin CM type systems were studied in [5][2].

Superintegrability of spin CM systems and of spin Ruijsenaars systems was established in [29]. In [31] the superintegrability of spin CM systems on homogeneous spaces was established. A family of superintegrable systems on moduli spaces of flat connections was constructed in [1]. This family includes systems studied in [5][2]. In these particular case the system is also Liouville integrable.

In this paper we will describe classical superintegrable system which we call spin Calogero-Moser (CM) chains. We call them spin CM chains because they combine features of many particle systems (as in CM systems) and of

spin chains. We distinguish two cases: a *periodic chain* and an *open chain*. The periodic case is the classical version of a quantum integrable system where joint eigenfunctions of quantum commuting Hamiltonians are trace functions, see [7]. In this case the spin part of the system reminds a spin chain with periodic boundary conditions. In case of rank 1 orbits for \mathfrak{sl}_n these systems are linearized versions of [5] and [2]. In the open case they are a classical version of quantum integrable systems constructed in [35][33]. For these systems the spin part of the system is similar to an open spin chain.

In both cases, i.e. in the periodic and in the open spin Calogero-Moser chains, the phases space is a stratified symplectic space [23], which, in some cases have only one stratum and becomes a symplectic manifold.

2. Recall that a *superintegrable system* is the structure on a symplectic manifold \mathcal{M} that consists of a Poisson manifold \mathcal{P} , a Poisson manifold \mathcal{B} with the trivial Poisson structure (i.e. zero Poisson tensor) and two surjective Poisson projections

(1)
$$\mathcal{M} \xrightarrow{p_1} \mathcal{P} \xrightarrow{p_2} \mathcal{B}$$

such that $\dim(\mathcal{M}) = \dim(\mathcal{P}) + \dim(\mathcal{B})$. For a superintegrable system a generic fiber of p_1 is an isotropic submanifolds of dimension $\dim(\mathcal{B})$ and connected components of a generic fiber of p_2 is a disjoint union of symplectic leaves of \mathcal{P} . For details see [26][30] and references therein. Here we adopt this notion to the case of stratified symplectic and Poisson spaces in which case p_1 and p_2 are Poisson mapping between stratifies spaces. In this paper the superintegrability means the balance of dimensions for the big stratum. How the system behave at smaller strata will be a subject of a separate publication. In the algebraic case, the appropriate setting is symplectic and Poisson stacks.

Let I be a Poisson commutative subalgebra of $A = C^{\infty}(\mathcal{M})$ that consists of functions which are constant on fibers of $p_2 \circ p_1$ (the pull-back of functions on \mathcal{B} to functions on \mathcal{M}) and J be the Poisson algebra of functions which are constant on fibers of p_1 (the pull-back of functions on \mathcal{P}). The condition on $(\mathcal{M}, \mathcal{P}, \mathcal{B})$ for being a superintegrable system is equivalent to the following condition on $I \subset J \subset A$. The Poisson algebra A has trivial center, $I \subset A$ is a Poisson commutative subalgebra, such that J, its centralizer in A maximal possible Gelfand-Kirillov dimension for the given Gelfand-Kirillov dimension of I.

The Hamiltonian dynamics generated by a function $H \in I$ is called *superintegrable*. Any function from J is constant along flow lines of the vector field generated by H and thus, it is an integral of motion for the Hamiltonian dynamics generated by H. This is why we call elements of the Poisson commutative subalgebra I Hamiltonians and elements of J conservation laws.

3. Throughout this paper G is a split real connected semisimple Lie algebra with finite center which admits a complexification, and $\Theta \in \text{Aut}(G)$

is a Cartan involution. We denote by $K = G^{\Theta}$ the closed subgroup of fixed points of Θ , which is connected and maximal compact. Let θ the corresponding Cartan involution¹ of \mathfrak{g} , and \mathfrak{k} the Lie algebra of K. The associated Cartan decomposition of \mathfrak{g} is $\mathfrak{g} = \mathfrak{k} \oplus \mathfrak{p}$, with \mathfrak{p} the (-1)-eigenspace of θ .

Let $\mathfrak{a} \subset \mathfrak{g}$ be maximally noncompact θ -stable Cartan subalgebra of \mathfrak{g} . Since \mathfrak{g} is split we have $\mathfrak{a} \subseteq \mathfrak{p}$. On the Lie group level, $A = \exp(\mathfrak{a}) \subset G$ is a maximal real split torus in G and $H := Z_G(A)$, the centraliser of A in G, is a Cartan subgroup in G containing A. The exponential map provides an isomorphism $\mathfrak{a} \xrightarrow{\sim} A$, whose inverse we denote by $\log : A \to \mathfrak{a}$.

Consider the root decomposition of $\mathfrak g$ with respect to the Cartan subalgebra $\mathfrak a,$

$$\mathfrak{g} = \mathfrak{a} \oplus igoplus_{lpha \in R} \mathfrak{g}_lpha$$

where $R \subset \mathfrak{a}^*$ is the root system of \mathfrak{g} relative to \mathfrak{a} . Choose $e_{\alpha} \in \mathfrak{g}_{\alpha}$ such that (2) $\theta(e_{\alpha}) = -e_{-\alpha}$

and $(e_{\alpha}, e_{-\alpha}) = 1$ for each $\alpha \in R$, and choose a subset $R_+ \subset R$ of positive roots. Let $W \subset \operatorname{GL}(\mathfrak{a}^*)$ be the Weyl group of R.

The Weyl group W is isomorphic to $N_G(A)/H$, where $N_G(A)$ is the normaliser of A in G. Denote by A_{req} the set of regular elements in $a \in A$,

$$A_{reg} := \{ a \in A \mid a_{\alpha} := e^{\alpha(\log(a))} \neq 1 \text{ for all } \alpha \in R \}.$$

It is the union of all the regular W-orbits in A. A fundamental domain for the W-action on A_{reg}^2 is the positive Weyl chamber

$$A_{+} = \{ a \in A \mid a_{\alpha} := e^{\alpha(\log(a))} > 1 \text{ for any } \alpha \in R_{+} \}.$$

Let $G' \subset G$ be the set of elements $g \in G$ which are *G*-conjugate to some element in A_{reg} . The inclusion $A_{reg} \hookrightarrow G'$ induces a bijection $A_{reg}/W \xrightarrow{\sim} G'/G$, with A_{reg}/W the set of *W*-orbits in A_{reg} and G'/G the set of *G*conjugacy classes in G'.

The Weyl group W is also isomorphic to $N_K(A)/M$, where $N_K(A) = N_G(A) \cap K$ and $M = H \cap K$ (note that M is a finite group since G is split). The inclusion map $A \hookrightarrow G$ induces an isomorphism $A/W \xrightarrow{\sim} K \setminus G/K$. We write $G_{reg} = KA_+K$ for the union of the double (K, K)-cosets intersecting A_{reg} .

4. The phase space of a periodic spin Calogero-Moser chain corresponding to a collection $\mathcal{O} = \{\mathcal{O}_1, \ldots, \mathcal{O}_n\}$ of coadjoint orbits $\mathcal{O}_i \subset \mathfrak{g}^*$ is the regular part of the symplectic leaf $\mathcal{S}(\mathcal{O})$ of the stratified Poisson space $T^*(G^{\times n})/G_n$, with the action of the gauge group $G_n = G^{\times n}$ the lift of a twisted conjugation action on $G^{\times n}$ (see section 1.1).³ Here we assume that

¹Recall that an involution $\theta : \mathfrak{g} \to \mathfrak{g}$ is a Cartan involution when the bilinear from $(-\theta(x), y)$ on \mathfrak{g} is positive definite. Here (\cdot, \cdot) is the Killing form.

²In case of $SL_n(\mathbb{R})$ one can take A to be the diagonal unimodular matrices with positive real entries, and A_{reg} consists of those with distinct diagonal entries.

³In this paper we assume that all quotients X/H are GIT quotients.

each of \mathcal{O}_i is non-trivial, i.e. $\mathcal{O}_i \neq \{0\}$ These symplectic leaves are obtained by the Hamiltonian reduction, as it is described in section 1.1. As a stratified symplectic space

$$\mathcal{S}(\mathcal{O}) \simeq \{ (x_1, \dots, x_n, g) \in \mathfrak{g}^{* \times n} \times G \mid x_1 - \mathrm{Ad}_{g^{-1}}^* x_n \in \mathcal{O}_1, \\ x_i - x_{i-1} \in \mathcal{O}_i, \ i = 2, \dots, n \} / G,$$

see section 1.3.⁴ Its regular part is defined as the intersection $\mathcal{S}(\mathcal{O})_{reg} = \mathcal{S}(\mathcal{O}) \cap (\mathfrak{g}^{* \times n} \times G')/G.^5$ The regular part has the following structure as a symplectic manifold, $\mathcal{S}(\mathcal{O})_{reg} \simeq (\nu_{\mathcal{O}}^{-1}(0)/H \times T^*A_{reg})/W$, where $\nu_{\mathcal{O}} : \mathcal{O}_1 \times \cdots \times \mathcal{O}_n \to \mathfrak{a}^*$ is the moment map for the diagonal coadjoint action of H on $\mathcal{O}_1 \times \cdots \times \mathcal{O}_n$ (see section 1.3).

Trivialization of T^*G by right translations gives an isomorphism $T^*(G^{\times n}) \simeq \mathfrak{g}^{*\times n} \times G^{\times n}$ and the Poisson projection $T^*(G^{\times n})/G_n \to (\mathfrak{g}^*/G)^{\times n}$, which is the projection to the cotangent directions followed by the quotienting with respect to the coadjoint action of $G^{\times n}$. Poisson commuting Hamiltonians of the periodic spin Calogero-Moser system are functions on $T^*(G^{\times n})/G_n$ which are constant on fibers of this Poisson projection. More precisely, tPoisson commuting Hamiltonians are such functions restricted to $\mathcal{S}(\mathcal{O})_{reg}$.

Consider the increasing set of natural numbers $2 = d_1 \leq \cdots \leq d_r$ with $r = \operatorname{rank}(\mathfrak{g})$ and $d_k - 1$ the exponents of \mathfrak{g} . Let c_{d_k} be the nonzero coadjoint invariant functions on \mathfrak{g}^* of degree d_k , known as Casimir functions. The function c_2 is the quadratic Casimir of \mathfrak{g} . Let $H_{d_k}^{(l)}$ be the function on $(\mathfrak{g}^*/G)^{\times n}$ which is c_{d_k} on the *l*-th factor and constant on all other factors.

Let us denote vectors in $\mathcal{O}_k \subset \mathfrak{g}^*$ by $\mu^{(k)}$, its Cartan component by $\mu_0^{(k)}$ and set $\mu_{\alpha}^{(k)} = \mu^{(k)}(e_{-\alpha})$ for $\alpha \in \mathbb{R}$. Denote (p, a) points on $T^*A \simeq \mathfrak{a}^* \times A$. Now let us describe quadratic Hamiltonians in terms of these variables.

The *n*-th quadratic Hamiltonian is the spin Calogero-Moser Hamiltonian. It has particularly simple form:

$$H_2^{(n)} = \frac{1}{2}(p,p) - \sum_{\alpha>0} \frac{\mu_{\alpha}\mu_{-\alpha}}{2\mathrm{sh}^2(q_{\alpha})}$$

where we used the parametrization $a_{\alpha} = e^{q_{\alpha}}$ (so $q_{\alpha} = \alpha(\log(a))$) and $\mu_{\alpha} = \mu_{\alpha}^{(1)} + \cdots + \mu_{\alpha}^{(n)}$, and (\cdot, \cdot) is the Euclidean form on \mathfrak{a}^* obtained by dualising the restriction of the Killing form of \mathfrak{g} to \mathfrak{a} .

⁴There are *n* such natural isomorphisms φ_j $(1 \leq j \leq n)$, see section 1.2. In the introduction we use φ_n .

⁵A better way to think about the periodic spin Calogero-Moser system for a real split simple Lie group G is to define $\mathcal{S}(\mathcal{O})_{\mathbb{C}}$ is complex algebraic setting and then to take the corresponding real slice. This will be addressed in another publication.

The differences $D_k = H_2^{(k)} - H_2^{(k-1)}$ for $1 < k \le n$ are classical analogs of topological Knizhnik-Zamolodchikov-Bernard differential operators,

(3)
$$D_k = (\mu_0^{(k)}, p) - \sum_{l=1}^{k-1} r_{lk} + \sum_{l=k+1}^n r_{kl}$$

where r_{kl} for $k \neq l$ is a classical version of the Felder's dynamical *r*-matrix [13],

(4)
$$r_{kl} = -\frac{1}{2}(\mu_0^{(k)}, \mu_0^{(l)}) + \sum_{\alpha} \frac{\mu_{-\alpha}^{(k)} \mu_{\alpha}^{(l)}}{a_{\alpha} - 1}$$

and \sum_{α} stands for the sum over all the roots $\alpha \in R$. This explicit form of D_k is derived in section 1.3.

The superintegrability of this system is described in section 1.5. The projection method for constructing solutions of equations of motion and angle variables are described in section 1.6.

One can choose G to be a the maximal compact real form of the complexification $G_{\mathbb{C}}$. In this case the integrable system is similar, but hyperbolic functions gets replaces by the trigonometric ones. The structure of the phase space is again a stratified symplectic space. The superintegrability of the quantum counterpart of such compact case is proven in [32].

5. The phase space of an open Calogero-Moser spin chain is the regular part of a symplectic leaf of the Poisson manifold $T^*(G^{\times n+1})/(K \times G^{\times n} \times K)$ where the action of the gauge group $K \times G^{\times n} \times K$ is described in section 2.3, and $K \subset G$ is as above. Such symplectic leaves are given by the Hamiltonian reduction. They are parametrized by collections of coadjoint orbits $\mathcal{O} = \{\mathcal{O}_{\ell}^K, \mathcal{O}_1, \ldots, \mathcal{O}_n, \mathcal{O}_r^K\}$ where $\mathcal{O}_i \subset \mathfrak{g}^*$ and $\mathcal{O}_{\ell,r}^K \subset \mathfrak{k}^* \subset \mathfrak{g}^*$ are coadjoint orbits. We assume that none of \mathcal{O}_i is trivial, i.e. $\mathcal{O}_i \neq \{0\}$.

We denote the corresponding symplectic leaf by $\mathcal{S}(\mathcal{O})$. It is a stratified symplectic space. Using Cartan decomposition G = KAK and a "gauge fixing fixing", we define the regular part $\mathcal{S}(\mathcal{O})_{reg}$ of $\mathcal{S}(\mathcal{O})$ as the stratum

$$S(\mathcal{O})_{reg} \simeq (T^*A_{reg} \times \mathcal{O}_1^K \times \mathcal{O}_1 \times \cdots \times \mathcal{O}_n \times \mathcal{O}_2^K))/N_K(A)$$

where on the right we have a natural product symplectic structure.

Similarly to the periodic case, quadratic Hamiltonians can be computed explicitly in terms of Cartan components $\mu_0^{(k)}$ and root coordinates $\mu_{\alpha}^{(k)}$ of vectors $\mu^{(k)} \in \mathcal{O}_k$, coordinates $\mu'_{[\alpha]}$, $\mu''_{[\alpha]}$ on \mathcal{O}_{ℓ}^K and \mathcal{O}_r^K respectively (in the basis elements $e_{[\alpha]} = e_{-\alpha} - e_{\alpha} \in \mathfrak{k} \subset \mathfrak{g}$ for $\alpha \in R_+$), and $(p, a) \in T^*A_{reg}$.

Assuming the gauge fixing ϕ_n (see section 2.3) we have

$$H_2^{(n)} = \frac{1}{2}(p,p) + \sum_{\alpha>0} \frac{(a_\alpha \mu'_{[\alpha]} + \mu''_{[\alpha]} + a_\alpha (\mu_\alpha - \mu_{-\alpha}))(a_\alpha^{-1}\mu'_{[\alpha]} + \mu''_{[\alpha]} + a_\alpha^{-1}(\mu_\alpha - \mu_{-\alpha}))}{(a_\alpha - a_{-\alpha})^2}$$

For other quadratic Hamiltonians the differences

$$D_k = H_2^{(k)} - H_2^{(k-1)} \qquad (1 \le k \le n)$$

are more interesting. They are classical analogs of boundary Knizhnik-Zamolodchikov-Bernard differential operators [35][33]. We have the following formula for D_k :

$$D_k = (\mu_0^{(k)}, p) - \sum_{l=1}^{k-1} (r_{lk} + r_{lk}^{\theta_l}) + (\sum_{\alpha} K_{\alpha} \mu_{-\alpha}^{(k)} - \kappa_k) + \sum_{l=k+1}^n (r_{kl} - r_{kl}^{\theta_k}).$$

Here r_{kl} for $k \neq l$ now is Felder dynamical *r*-matrix rescaled in $a \in A_{reg}$,

(5)
$$r_{kl} = -\frac{1}{2}(\mu_0^{(k)}, \mu_0^{(l)}) + \sum_{\alpha} \frac{\mu_{-\alpha}^{(k)} \mu_{\alpha}^{(l)}}{a_{\alpha}^2 - 1}$$

 θ_k is the transpose of the Cartan involution acting on $\mu^{(k)}$,

$$\kappa_k = \frac{1}{2}(\mu_0^{(k)}, \mu_0^{(k)}) + \sum_{\alpha} \frac{(\mu_{\alpha}^{(k)})^2}{1 - a_{\alpha}^2}$$

and

(6)
$$K_{\alpha} = \frac{a_{\alpha}\mu'_{[\alpha]} + \mu''_{[\alpha]}}{a_{\alpha} - a_{\alpha}^{-1}}.$$

The differences $D_k = H_2^{(k)} - H_2^{(k-1)}$ are classical analogs of boundary KZB operators derived in [35][33]. The superintegrability of open spin CM chains is proven in section 2.6. The projection method for solving equations of motion and angle coordinates are described in section 2.7.

6. The structure of the paper is as follows. In section 1 we construct periodic spin CM chains by the Hamiltonian reduction and prove the superintegrability. In section 1.1 we describe the phase space of such a system. In sections 1.2, 1.3 we describe the regular part of the phase space. Hamiltonians of a periodic spin CM chain, restricted to the regular part of the phase space are described in section 1.4. The superintegrability of a periodic spin CM chain is proven in section 1.5. In section 1.6 solutions to equations of motion are described algebraically by the projection method, and angle variables are described. In section 2 we focus on open spin CM chains. In section 2.1 we describe phase spaces. In section 2.3, 2.4 we describe the regular part of the phase space. Hamiltonians of an open spin CM chain, restricted to the regular part of the phase space are described in section 2.5. The superintegrability of an open spin CM chain is proven in section 2.6. In section 2.7 solutions to equations of motion are described algebraically by the projection method, and angle variables are described. In the conclusion (section 3) we discuss some open problems and describe in details periodic CM spin chain for SL_N with orbits of rank 1. In Appendix A we compare our symplectic leaves with the ones from [31].

Throughout this paper we will focus on split real semisimple Lie groups. However, since all constructions are algebraic they extend (with appropriate modifications) to the complex algebraic case. The non-split real case will be the subject of a separate publication (see [33] for the quantum case). Another important real case is when G is compact, which can be deduced from the complex algebraic case by restriction to a compact real form. The structure of phase spaces as stratified symplectic spaces will be explored further in [6].

Acknowledgments. This paper was started as a joint project with Jasper Stokman. The author is grateful to Jasper for many discussions and for the collaboration on this paper. He also would like to thank Vladimir Fock, Eva Miranda and Hessel Postuma for important discussions and remarks and to Zhuo Chen, Kai Jiang and Husileng Xiao for discussions on stratified symplectic spaces. N.R. want to thank ITS-ETH for the hospitality, where the bulk of this work was completed. The work of N.R. was supported by the NSF grant DMS-1902226, by the grant RFBR No. 18-01-00916, by the Collaboration Grant "Categorical Symmetries" from the Simons Foundation, by the grant BMSTC and ACZSP (Grant no. Z221100002722017) and by the Changjiang fund.

1. Periodic spin Calogero-Moser chains

1.1. The phase space as the Hamiltonian reduction. Here we will describe the phase space of a periodic spin Calogero-Moser chain as a Hamiltonian reduction of $T^*(G^{\times n})$. Let us start with the description of these symplectic spaces.

Consider the manifold $T^*(G^{\times n})$ with the standard symplectic structure. The cotangent bundle over a Lie group can be trivialized by right translations, which gives an isomorphism of vector bundles

$$T^*(G^{\times n}) \simeq (T^*G)^{\times n} \simeq \mathfrak{g}^{*\times n} \times G^{\times n}$$

We will choose this trivialization throughout the paper.

The Lie group $G_n := G^{\times n}$ acts naturally on itself by left and right translations. Lifting these actions to $T^*(G^{\times n})$, after the trivialization of the cotangent bundle, we can write the action by left translations as:

$$h_L(x,g) = (Ad_{h_1}^*(x_1), Ad_{h_2}^*(x_2), \dots, Ad_{h_n}^*(x_n), h_1g_1, h_2g_2, \dots, h_ng_n)$$

and the action by right translations as

 $h_R(x,g) = (x_1, \dots, x_n, g_1 h_1^{-1}, \dots, g_n h_n^{-1})$

Both these actions are Hamiltonian with moment maps

$$\mu_L(x,g) = (x_1, x_2, \dots, x_n)$$

and

$$\mu_R(x,g) = (-Ad_{g_1^{-1}}^*(x_1), \dots, -Ad_{g_n^{-1}}^*(x_n))$$

respectively.

N. RESHETIKHIN

Actions by left and right translations can be twisted by permutations. In particular, we can twist the action by left translations by a cyclic permutation. Combining the twisted left action with the non-twisted right action we obtain the "gauge action" of G_n on $G^{\times n^6}$

$$h(g_1, \dots, g_n) = (h_1 g_1 h_2^{-1}, h_2 g_2 h_3^{-1}, \dots, h_n g_n h_1^{-1})$$

Lifting the twisted conjugation action of G_n on $G^{\times n}$ to $T^*(G^{\times n})$ we obtain the "gauge action" on the cotangent bundle:

(7)
$$h(x,g) = (Ad_{h_1}^*(x_1), Ad_{h_2}^*(x_2), \dots, Ad_{h_n}^*(x_n), h_1g_1h_2^{-1}, h_2g_2h_3^{-1}, \dots, h_ng_nh_1^{-1})$$

Because this is the diagonal action for two Hamiltonian actions, the gauge action is also Hamiltonian with the moment map $\mu: T^*(G^{\times n}) \to \mathfrak{g}^{*\times n}$:

(8)
$$\mu(x,g) = \mu_L(x,g) + \mu_R^{tw}(x,g) = (x_1 - Ad_{g_n^{-1}}^*(x_n), x_2 - Ad_{g_1^{-1}}^*(x_1), \dots, x_n - Ad_{g_{n-1}^{-1}}^*(x_{n-1}))$$

where μ_B^{tw} is the right moment map, twisted by cyclic permutation.

Because the gauge action (7) of G_n is Hamiltonian, the quotient space $T^*(G^{\times n})/G_n$ is a Poisson space.⁷ Symplectic leaves of $T^*(G^{\times n})/G_n$ are given by the Hamiltonian reduction with respect to the moment map (8). Let $\mathcal{O}_1, \ldots, \mathcal{O}_n$ be coadjoint orbits in \mathfrak{g}^* , then the corresponding symplectic leaf in $T^*(G^{\times n})/G_n$ is

(9)
$$\mathcal{S}(\mathcal{O}) = \mu^{-1}(\mathcal{O}_1 \times \dots \times \mathcal{O}_n)/G_n \\ = \{(x,g) \in \mathfrak{g}^{* \times n} \times G^{\times n} | x_i - Ad_{g_{i-1}}^*(x_{i-1}) \in \mathcal{O}_i\}/G_n$$

where G_n acts by the gauge transformations (7) and the indices *i* should be taken modulo *n*.

On each of these symplectic leaves we will construct a superintegrable system which we will call a *periodic spin Calogero-Moser chain*.

1.2. The gauge fixing. Let us fix $i \in 1, ..., n$ and $g = (g_1, ..., g_n) \in G^{\times n}$. Let $h \in G_n$ such that

$$h_j = \begin{cases} h_i g_{i-1}^{-1} \cdots g_{j+1}^{-1} g_j^{-1} & \text{for } 1 \le j < i, \\ h_i g_{i-1}^{-1} \cdots g_2^{-1} g_1^{-1} g_n^{-1} \cdots g_{j+1}^{-1} g_j^{-1} & \text{for } i < j \le n. \end{cases}$$

Denote such element of G_n by h_q (we suppress the dependence on *i*).

It is easy to check that the gauge transformation of $g = (g_1, \ldots, g_n)$ by the element h_g brings it to $(1, \ldots, 1, h_i(g_ig_{i+1}\cdots g_ng_1g_2, \ldots, g_{i-1})h_i^{-1}, 1, \ldots, 1)$,

 $^{^6 \}mathrm{One}$ can twist both left and right actions by a permutation. This leads to other superintegrable systems.

⁷This space is singular. Having in mind classical-quantum correspondence we need the algebra of functions on $T^*G^{\times n}/G_n$. Thus, by the quotient space we will always mean the GIT quotient. By definition, functions on $T^*(G^{\times n})/G_n$ are G_n -invariant functions on $T^*(G^{\times n})$.

with the *i*th-entry being the nontrivial entry. This identifies the G_n gauge orbit through $g = (g_1, \ldots, g_n)$ with the *G*-conjugation orbit through $g_1 \cdots g_n$. It thus gives an (*i*-independent) isomorphism

$$G^{\times n}/G_n \xrightarrow{\sim} G/G,$$

where G/G denotes the set of conjugacy classes in G. On the cotangent bundles the gauge fixing with gives the isomorphism $\varphi_i : (\mathfrak{g}^{*\times n} \times G^{\times n})/G_n \xrightarrow{\sim} (\mathfrak{g}^{*\times n} \times G)/G$ mapping the G_n -orbit $G_n(x,g)$ through $(x,g) \in \mathfrak{g}^{*\times n} \times G^{\times n}$ to the G-orbit through

$$(Ad_{g_i^{-1}\cdots g_1^{-1}}^{*}(x_1), \dots, Ad_{g_i^{-1}}^{*}(x_i), Ad_{g_i^{-1}\cdots g_1^{-1}g_n^{-1}\cdots g_{i+1}^{-1}}^{*}(x_{i+1}), \dots, Ad_{g_i^{-1}\cdots g_1^{-1}g_n^{-1}}^{*}(x_n), g_{i+1}\cdots g_ng_1\cdots g_i)$$

(for i = n this should be read as $\left(Ad_{g_n^{-1}\cdots g_1^{-1}}^*(x_1), \ldots, Ad_{g_n^{-1}}^*(x_n), g_1\cdots g_n\right)$). Here G is acting diagonally on $\mathfrak{g}^{*\times n} \times G$ via the coadjoint action on \mathfrak{g}^* and the conjugation action on G. From now on we will work with the isomorphism φ_n .

1.3. The regular part of the phase space. The image of the symplectic leaf $\mathcal{S}(\mathcal{O})$ under the isomorphism φ_n is

$$\mathcal{S}(\mathcal{O}) = \{ (z_1, \dots, z_n, g) \in \mathfrak{g}^{* \times n} \times G \mid z_1 - Ad_{g^{-1}}^* z_n \in \mathcal{O}_1, \\ z_i - z_{i-1} \in \mathcal{O}_i, \quad i = 2, \dots, n \} / G.$$

Define the regular part $\mathcal{S}(\mathcal{O})_{reg} \subset \mathcal{S}(\mathcal{O})$ of the phase space as $\mathcal{S}(\mathcal{O}) \cap (\mathfrak{g}^{*\times n} \times G')/G$. On $\mathcal{S}(\mathcal{O})_{reg}$ we can choose a representative where g is in the regular part A_{reg} of the real split torus A in G: $g = bab^{-1}$, $z_i = \mathrm{Ad}_b^* x^{(i)}$ with $a \in A_{reg}$. Then we have

$$\mathcal{S}(\mathcal{O})_{reg} = \{ (x^{(1)}, \dots, x^{(n)}, a) \in \mathfrak{g}^{* \times n} \times A_{reg} \mid x^{(1)} - Ad^*_{a^{-1}} x^{(n)} \in \mathcal{O}_1, \\ x^{(i)} - x^{(i-1)} \in \mathcal{O}_i, \\ i = 2, \dots, n \} / N_G(A).$$

Identify $\mathfrak{g}^* \simeq \mathfrak{g}$ and $\mathfrak{a}^* \simeq \mathfrak{a}$ via the Killing form of \mathfrak{g} . The element $y \in \mathfrak{g}^*$ then corresponds to $y_0 + \sum_{\alpha} y_{\alpha} e_{\alpha}$, where y_0 is the element in \mathfrak{a} corresponding to $y|_{\mathfrak{a}}$ and $y_{\alpha} = y(e_{-\alpha})$. Let $\mu^{(j)} \in \mathcal{O}_j$ be vectors $\mu^{(1)} = x^{(1)} - Ad_{\mathfrak{a}^{-1}}^* x^{(n)}$ and $\mu^{(i)} = x^{(i)} - x^{(i-1)}$ for $i = 2, \ldots, n$. For coordinates $x_{\alpha}^{(i)}$ and $\mu_{\alpha}^{(i)}$ of vectors $x^{(i)}$ and $\mu^{(i)}$ we then have

$$x_{\alpha}^{(1)} - a_{\alpha}^{-1} x_{\alpha}^{(n)} = \mu_{\alpha}^{(1)}, \qquad x_{\alpha}^{(i)} - x_{\alpha}^{(i-1)} = \mu_{\alpha}^{(i)}, \qquad i = 2, \dots, n$$

For the Cartan components we have

$$x_0^{(i)} - x_0^{(i-1)} = \mu_0^{(i)}, \qquad i = 1, \dots, n_s$$

with the index i taken to be modulo n.

Solving these equations for $x^{(i)}$ we have

(10)
$$x_{\alpha}^{(i)} = \frac{a_{\alpha}(\mu_{\alpha}^{(1)} + \mu_{\alpha}^{(2)} + \dots + \mu_{\alpha}^{(i)}) + \mu_{\alpha}^{(i+1)} + \mu^{(i+2)} + \dots + \mu_{\alpha}^{(n)}}{a_{\alpha} - 1},$$
$$x_{0}^{(i)} = x_{0}^{(1)} + \mu_{0}^{(2)} + \dots + \mu_{0}^{(i)} = x_{0}^{(n)} - \mu_{0}^{(n)} - \dots - \mu_{0}^{(i+1)}$$

and we have the constraint

(11)
$$\mu_0^{(1)} + \dots + \mu_0^{(n)} = 0.$$

This gives an isomorphism

(12)
$$\mathcal{S}(\mathcal{O})_{reg} \xrightarrow{\sim} \left(\nu_{\mathcal{O}}^{-1}(0)/H \times T^*A_{reg}\right)/W$$

which preserves the natural symplectic structures, where $\nu_{\mathcal{O}} : \mathcal{O}_1 \times \cdots \times \mathcal{O}_n \to \mathfrak{a}^*$ is the moment map $(\mu^{(1)}, \ldots, \mu^{(n)}) \mapsto (\mu^{(1)} + \cdots + \mu^{(n)})|_{\mathfrak{a}}$ for the diagonal action of H on the product $\mathcal{O}_1 \times \cdots \times \mathcal{O}_n$ of coadjoint orbits, and $W = N_G(A)/H$ acts diagonally on $\nu_{\mathcal{O}}^{-1}(0)/H \times T^*A_{reg}$. The isomorphism (12) maps the $N_G(A)$ -orbit through $(x^{(1)}, \ldots, x^{(n)}, a)$ to the Worbit through $(H(x^{(1)} - Ad^*_{a^{-1}}x^{(n)}, x^{(2)} - x^{(1)}, \ldots, x^{(n)} - x^{(n-1)}), x_0^{(n)}, a)$, where we used the trivialisation $T^*A_{reg} \simeq \mathfrak{a} \times A_{reg}$. The inverse maps the W-orbit through $(H(\mu^{(1)}, \ldots, \mu^{(n)}), p, a)$ to the $N_G(A)$ -orbit through $(x^{(1)}, \ldots, x^{(n)}, a)$, with

(13)
$$x^{(i)} = p - \mu_0^{(n)} - \dots - \mu_0^{(i+1)} + \sum_{\alpha} \left(\frac{a_{\alpha}(\mu_{\alpha}^{(1)} + \dots + \mu_{\alpha}^{(i)}) + \mu_{\alpha}^{(i+1)} + \dots + \mu_{\alpha}^{(n)}}{a_{\alpha} - 1} \right) e_{\alpha},$$

where we use the identification $\mathfrak{g} \simeq \mathfrak{g}^*$ via the Killing form.

1.4. Hamiltonians of a periodic spin CM chain. After the trivialization of the cotangent bundle by translations, we have a natural projection:

(14)
$$T^*(G^{\times n}) \simeq \mathfrak{g}^{*\times n} \times G^{\times n} \to \mathfrak{g}^{*\times n}$$

which is simply the projection to the first factor. This projection depends on the trivialization. In this paper we alway assume that we use the trivialization by right translations. However, the corresponding projection of quotient spaces

(15)
$$T^*(G^{\times n})/G_n \to (\mathfrak{g}^*/G)^{\times n}$$

does not depend on the trivialization and in this sense is canonical.

The projection (15) is Poisson⁸ with the trivial Poisson structure on $(\mathfrak{g}^*/G)^{\times n}$. Thus the $G^{\times n}$ -invariant functions on $\mathfrak{g}^{*\times n}$ give rise to a Poisson commutative subalgebra in the algebra of functions on $T^*(G^{\times n})/G_n$. The restriction of these functions to the symplectic leaf $\mathcal{S}(\mathcal{O})$ gives the algebra of

⁸One of the reasons for this is that the equation (14) is the moment map for the left diagonal action of $G^{\times n}$ on the cotangent bundle.

Poisson commuting functions on it. This is the subalgebra of *Hamiltonians* of the periodic spin Calogero-Moser chain.

Now let us describe the restriction of the Hamiltonians corresponding to quadratic Casimir functions

$$H_2^{(k)}(x,g) = \frac{1}{2} \left(x^{(k)}, x^{(k)} \right) = \frac{1}{2} \left(x_0^{(k)}, x_0^{(k)} \right) + \sum_{\alpha > 0} x_\alpha^{(k)} x_{-\alpha}^{(k)} \qquad (1 \le k \le n)$$

to the regular part of $\mathcal{S}(\mathcal{O})$, where $x = (x^{(1)}, \ldots, x^{(n)}) \in \mathfrak{g}^{* \times n}$ and $g \in G^{\times n}$. Consider the functions

$$D_k = H_2^{(k)} - H_2^{(k-1)} \qquad (1 < k \le n)$$

which we call Knizhnik-Zamolodchikov-Bernard (KZB) Hamiltonians⁹.

THEOREM 1. The restriction of the KZB Hamiltonians to $\mathcal{S}(\mathcal{O})_{reg}$ can be written as

(16)
$$D_k = (\mu_0^{(k)}, p) - \sum_{l=1}^{k-1} r_{lk} + \sum_{l=k+1}^n r_{kl}$$

where r_{kl} for $k \neq l$ is the classical version of the Felder's dynamical r-matrix [13]:

(17)
$$r_{kl} = -\frac{1}{2}(\mu_0^{(k)}, \mu_0^{(l)}) + \sum_{\alpha} \frac{\mu_{-\alpha}^{(k)} \mu_{\alpha}^{(l)}}{a_{\alpha} - 1}$$

REMARK 1. Note that (17) can also be written as

$$r_{kl} = -\frac{1}{2}(\mu_0^{(k)}, \mu_0^{(l)}) + \sum_{\alpha>0} \frac{\mu_{-\alpha}^{(k)} \mu_{\alpha}^{(l)}}{a_{\alpha} - 1} - \sum_{\alpha>0} \frac{a_{\alpha} \mu_{\alpha}^{(k)} \mu_{-\alpha}^{(l)}}{a_{\alpha} - 1}.$$

PROOF. We need to show that formula (16) gives the expression of D_k in terms of the coordinates on $\mathcal{S}(\mathcal{O})_{reg}$ of $(\nu_{\mathcal{O}}^{-1}(0)/H \times T^*A_{reg})/W$, obtained from the isomorphism (12). In particular, let $(H(\mu^{(1)}, \ldots, \mu^{(n)}), p, a) \in \nu_{\mathcal{O}}^{-1}(0)/H \times \mathfrak{a}^* \times A_{reg}$ and let $((x^{(1)}, \ldots, x^{(n)}), (1, \ldots, 1, a))$ be the corresponding point in $\mathfrak{g}^{*\times n} \times G^{\times n}$, with $x^{(i)}$ given by (13). Taking into account the relation $x^{(k)} - x^{(k-1)} = \mu^{(k)}$ between the $x^{(i)}$ and the $\mu^{(j)}$ we have

$$D_{k} = (\mu^{(k)}, x^{(k-1)}) + \frac{1}{2} (\mu^{(k)}, \mu^{(k)})$$

= $(\mu_{0}^{(k)}, x_{0}^{(k-1)} + \frac{1}{2} \mu_{0}^{(k)}) + \sum_{\alpha} x_{\alpha}^{(k-1)} \mu_{-\alpha}^{(k)} + \sum_{\alpha>0} \mu_{\alpha}^{(k)} \mu_{-\alpha}^{(k)}.$

⁹The proper name would be *constant Knizhnik-Zamolodchikov-Bernard Hamiltonians* emphasizing the fact that they are related to finite dimensional simple Lie algebras, not to the affine Kac-Moody algebras. See for example references [13][8][9][34].

Substitute here the expression (10) for $x_{\alpha}^{(k-1)}$ in terms of the $\mu_{\alpha}^{(j)}$:

$$D_k = \left(\mu_0^{(k)}, x_0^{(k-1)} + \frac{1}{2}\mu_0^{(k)}\right) + \sum_{l=1}^{k-1} \sum_{\alpha} \frac{a_{\alpha}\mu_{\alpha}^{(l)}\mu_{-\alpha}^{(k)}}{a_{\alpha} - 1} + \sum_{l=k+1}^n \sum_{\alpha} \frac{\mu_{\alpha}^{(l)}\mu_{-\alpha}^{(k)}}{a_{\alpha} - 1}.$$

From here using the identities

$$\sum_{\alpha} \frac{a_{\alpha} \mu_{\alpha}^{(l)} \mu_{-\alpha}^{(k)}}{a_{\alpha} - 1} = -r_{lk} - \frac{1}{2} (\mu_0^{(k)}, \mu_0^{(l)})$$

and

$$\sum_{\alpha} \frac{\mu_{\alpha}^{(l)} \mu_{-\alpha}^{(k)}}{a_{\alpha} - 1} = r_{kl} + \frac{1}{2} (\mu_0^{(k)}, \mu_0^{(l)})$$

we conclude

$$D_k = \left(\mu_0^{(k)}, x_0^{(k-1)} + \frac{1}{2}\sum_{l=k}^n \mu_0^{(l)} - \frac{1}{2}\sum_{l=1}^{k-1} \mu_0^{(l)}\right) - \sum_{l=1}^{k-1} r_{lk} + \sum_{l=k+1}^n r_{kl}.$$

Using $x_0^{(k-1)} = p - \mu_0^{(n)} - \dots - \mu_0^{(k)}$ (see (13)) and the constraint (11) we obtain (16).

A particularly simple expression has the quadratic Hamiltonian $H_2^{(n)}$ on $\mathcal{S}(\mathcal{O})_{reg}$,

$$H_2^{(n)} = \frac{1}{2}(p,p) + \sum_{\alpha} \frac{\mu_{\alpha}\mu_{-\alpha}}{(1-a_{\alpha})(1-a_{\alpha}^{-1})}.$$

Here $\mu_{\alpha} = \mu_{\alpha}^{(1)} + \cdots + \mu_{\alpha}^{(n)}$. Setting $q_{\alpha} = \alpha(\log(a))$ this formula becomes a familiar formula for the spin Calogero-Moser Hamiltonian,

(18)
$$H_2^{(n)} = \frac{1}{2}(p,p) - \sum_{\alpha>0} \frac{\mu_{\alpha}\mu_{-\alpha}}{2sh^2(q_{\alpha})}.$$

Note that the periodic spin CM chain is the classical version of the dynamical Knizhnik-Zamolodchikov equation from [8][9].

1.5. Periodic spin Calogero-Moser chain as a superintegrable system. Now let us establish the superintegrability of the periodic spin CM chain. For this we should construct an intermediate Poisson manifold and projections as in [26][30].

Observe that we have natural Poisson projections:

(19)
$$T^{*}(G^{\times n})/G_{n} \xrightarrow{p_{1}} \mathcal{P}_{n} \xrightarrow{p_{2}} \mathcal{B}_{n}$$

Firstly, $\mathcal{P}_{n} = (\mathfrak{g}^{*\times n} \times_{(\mathfrak{g}^{*}/G)^{\times n}} \mathfrak{g}^{*\times n})/G_{n}$ with
 $\mathfrak{g}^{*\times n} \times_{(\mathfrak{g}^{*}/G)^{\times n}} \mathfrak{g}^{*\times n} := \{(x, y) \in \mathfrak{g}^{*\times n} \times \mathfrak{g}^{*\times n} | Gy_{i} = -Gx_{i-1}\},$

where Gz is the coadjoint orbit through $z \in \mathfrak{g}^*$ and the indices *i* are taken modulo *n*, and G_n is acting by

(20)
$$g(x,y) := (Ad_{g_1}^*(x_1), \dots, Ad_{g_n}^*(x_n), Ad_{g_1}^*(y_1), \dots, Ad_{g_n}^*(x_n)).$$

The map p_1 is the map induced from the G_n -equivariant map $\mu_L \times \mu_R^{tw}$. Explicitly, the mapping p_1 acts as

(21)
$$p_{1}: G_{n}(x,g) \mapsto G_{n}(\mu_{L}(x,g), \mu_{R}^{tw}(x,g)) = G_{n}(x_{1}, x_{2}, \dots, x_{n}, -Ad_{g_{n}^{-1}}^{*}(x_{n}), -Ad_{g_{1}^{-1}}(x_{1}), \dots, -Ad_{g_{n-1}^{-1}}(x_{n-1})).$$

Secondly,

$$\mathcal{B}_n = \left(\mathfrak{g}^*/G\right)^{\times n}$$

and the map p_2 is the projection to the first factor.

Restricting projection p_1 to the symplectic leaf $\mathcal{S}(\mathcal{O})$ (see (9)), we obtain the surjective Poisson projection

$$p_{1,\mathcal{O}}: \mathcal{S}(\mathcal{O}) \to \mathcal{P}(\mathcal{O})$$

where

$$\mathcal{P}(\mathcal{O}) = \{(x, y) \in \mathfrak{g}^{* \times n} \times_{(\mathfrak{g}^*/G)^{\times n}} \mathfrak{g}^{* \times n} \mid x_i + y_i \in \mathcal{O}_i\}/G_n \subset \mathcal{P}_n$$

with the G_n -action described by (20).

Restricting the second projection p_2 to $\mathcal{P}(\mathcal{O})$ we have the Poisson projection

 $p_{2,\mathcal{O}}: \mathcal{P}(\mathcal{O}) \to \mathcal{B}(\mathcal{O}) \subset \mathcal{B}_n, \qquad G_n(x,y) \mapsto (Gx_1, \dots, Gx_n)$

where $\mathcal{B}(\mathcal{O})$ is the image of $p_{2,\mathcal{O}}$. It can be explicitly described as

$$\mathcal{B}(\mathcal{O}) = \{ (\mathcal{O}^{(1)}, \dots, \mathcal{O}^{(n)}) \in (\mathfrak{g}^*/G)^{\times n} \mid \mathcal{O}_i \subseteq \mathcal{O}^{(i)} - \mathcal{O}^{(i-1)} \},\$$

with the indices i taken modulo n.

LEMMA 1. The dimension of $\mathcal{B}(\mathcal{O})$ is nr where r is the rank of the Lie algebra \mathfrak{g} .

PROOF. Let $\mathfrak{h}_{\geq 0}^*$ be the positive Weyl chamber in the dual space to Cartan subalgebra of \mathfrak{g} . For each generic orbit \mathcal{O} there is a unique representative $y \in \mathcal{O} \cup \mathfrak{h}^* > 0$. Let x_1 be such representative of $\mathcal{O}^{(1)}$. Let us describe orbits $\mathcal{O}^{(2)}$ such that $x_1 + y_1 \in \mathcal{O}^{(2)}$ for some $y_1 \in \mathcal{O}_1$. Assume that $\mathcal{O}^{(1)}$ is very large, i.e. $||x_1|| >> 1$. Because \mathcal{O}_1 is compact $||y_1|| < C_1$ for some constant C_1 determined by the orbit \mathcal{O}_1 . Let $c_k^{(i)}$ be the value of k-th Casimir function on the orbit $\mathcal{O}^{(i)}$.

For k-th Casimir function $c_k^{(2)}$ we have:

$$c_k^{(2)} = c_k(x_1 + y_1) = c_k^{(1)} + \sum_{i=1}^r \frac{\partial c_k(h)}{\partial h_i}|_{h=x_1}(y_1)_i + O(y^2)$$

Because the matrix $\frac{\partial c_k(h)}{\partial h_i}$ is nondegerate for generic h, possible values of the Euclidean vector with components $c_k^{(2)}$ span an r-dimensional neighborhood of $\{c_k^{(1)}\}$.

Repeating this argument for each $\mathcal{O}^{(i)}$ we conclude that each of \mathcal{O}_i is non-zero, $\dim(\mathcal{B}(\mathcal{O})) = nr$.

Now let us describe the fiber $\mathcal{P}(\mathcal{O}; \mathcal{O}^{(1)}, \dots, \mathcal{O}^{(n)})$ of $p_{2,\mathcal{O}}$ over $(\mathcal{O}^{(1)}, \dots, \mathcal{O}^{(n)}) \in \mathcal{B}(\mathcal{O})$:

$$\mathcal{P}(\mathcal{O}; \mathcal{O}^{(1)}, \dots, \mathcal{O}^{(n)})$$

$$= \{(x, y) \in \mathfrak{g}^{* \times n} \times \mathfrak{g}^{* \times n} | x_i + y_i \in \mathcal{O}_i, \quad x_i \in \mathcal{O}^{(i)}, y_i \in -\mathcal{O}^{(i-1)}\}/G_n$$

$$= \prod_{i=1}^n \{(x_i, y_i) \in \mathcal{O}^{(i)} \times -\mathcal{O}^{(i-1)} \mid x_i + y_i \in \mathcal{O}_i\}/G$$

with the index *i* taken modulo *n* and with *G* acting by the diagonal coadjoint action on $\mathcal{O}^{(i)} \times -\mathcal{O}^{(i-1)}$. Set

(22)
$$\mathcal{M}(\mathcal{O}^{(1)}, \mathcal{O}^{(2)}, \mathcal{O}^{(3)}) = \{(x, y, z) \in \mathcal{O}^{(1)} \times \mathcal{O}^{(2)} \times \mathcal{O}^{(3)} \mid x + y + z = 0\}/G,$$

with G acting by the diagonal coadjoint action. Then

$$\{ (x_i, y_i) \in \mathcal{O}^{(i)} \times -\mathcal{O}^{(i-1)} \mid x_i + y_i \in \mathcal{O}_i \} / G \xrightarrow{\sim} \mathcal{M}(-\mathcal{O}^{(i)}, \mathcal{O}^{(i-1)}, \mathcal{O}_i)$$

$$G(x_i, y_i) \mapsto G(-x_i, -y_i, x_i + y_i),$$

and hence we conclude that

(24)
$$\mathcal{P}(\mathcal{O}; \mathcal{O}^{(1)}, \dots, \mathcal{O}^{(n)}) \simeq \prod_{i=1}^{n} \mathcal{M}(-\mathcal{O}^{(i)}, \mathcal{O}^{(i-1)}, \mathcal{O}_{i}),$$

with the index i taken modulo n.

LEMMA 2. Let $\mathcal{O}^{(1)}$, $\mathcal{O}^{(2)}$ be generic, sufficiently large coadjoint orbits and $\mathcal{O}^{(3)} \neq 0$, then

$$\dim(\mathcal{M}(\mathcal{O}^{(1)}, \mathcal{O}^{(2)}, \mathcal{O}^{(3)})) = \dim(\mathcal{O}^{(3)}) - 2r.$$

PROOF. Let $x \in \mathcal{O}^{(1)}$ be the unique representative which lies in the positive Weyl chamber. Assume this orbit is "big", i.e. ||x|| >> 1. The condition x + y + z = 0 for $y \in \mathcal{O}^{(2)}$ and $z \in \mathcal{O}^{(3)}$ for a large orbit $\mathcal{O}^{(2)}$ means that we have r constraints $c_k(-x-y) = c_k^{(3)}$ on y. For large orbits $\mathcal{O}^{(1)}$ and $\mathcal{O}^{(2)}$ these constraints are independent. Taking into account that we are quotienting by H we have $\dim(\mathcal{M}(\mathcal{O}^{(1)}, \mathcal{O}^{(2)}, \mathcal{O}^{(3)})) = \dim(\mathcal{O}^{(3)}) - 2r$. \Box

COROLLARY 1. Thus the dimension of the fiber is $\dim(\mathcal{P}(\mathcal{O}; \mathcal{O}^{(1)}, \ldots, \mathcal{O}^{(n)})) = \sum_{i=1}^{n} \dim(\mathcal{O}_i) - 2nr.$

Each of the factors in (24) is the Hamiltonian reduction of the product of the three coadjoint orbits relative to the moment map of the diagonal coadjoint *G*-action, and therefore carries a natural symplectic structure. Moduli spaces $\mathcal{M}(\mathcal{O}^{(1)}, \mathcal{O}^{(2)}, \mathcal{O}^{(3)})$ and therefore fibers of $p_{2,\mathcal{O}}$ are stratified symplectic spaces.

THEOREM 2. The Hamiltonian system generated by any Hamiltonian for the periodic spin CM chain described in section 1.4 is superintegrable with the superintegrable structure described by the Poisson maps

$$\mathcal{S}(\mathcal{O}) \xrightarrow{p_{1,\mathcal{O}}} \mathcal{P}(\mathcal{O}) \xrightarrow{p_{2,\mathcal{O}}} \mathcal{B}(\mathcal{O})$$

as introduced earlier in this section.

Here, as everywhere above, we assume that $\mathcal{O}_i \neq \{0\}$ for each $i = 1, \ldots, n$.

PROOF. For $G_n(x,y) \in \mathcal{P}(\mathcal{O})$ let $\tilde{g}_i \in G$ such that $y_{i+1} = -Ad^*_{\tilde{g}_i^{-1}}(x_i)$. Then

$$p_{1,\mathcal{O}}^{-1}(G_n(x,y)) = \{G_n(x,g) \in \mathcal{S}(\mathcal{O}) \mid g_i \in \widetilde{g}_i Z_G(y_{i+1})\}$$

(index *i* taken modulo *n*) which, generically, is isotropic and of dimension $nr = \dim(\mathcal{B}(\mathcal{O}))$.

It remains to check the balance of dimensions. It follows from (12) that

$$\dim(\mathcal{S}(\mathcal{O})) = \sum_{i=1}^{n} \dim(\mathcal{O}_i).$$

By Remark 2 we have, for generic $(\mathcal{O}^{(1)}, \ldots, \mathcal{O}^{(n)}) \in (\mathfrak{g}^*/G)^{\times n}$,

$$\dim(\mathcal{P}(\mathcal{O})) = \dim(\mathcal{B}(\mathcal{O})) + \dim(\mathcal{P}(\mathcal{O}; \mathcal{O}^{(1)}, \dots, \mathcal{O}^{(n)}))$$
$$= \dim(\mathcal{B}(\mathcal{O})) + \sum_{i=1}^{n} \dim(\mathcal{O}_i) - 2nr.$$

Then

$$\dim(\mathcal{P}(\mathcal{O})) + \dim(\mathcal{B}(\mathcal{O})) = \sum_{i=1}^{n} \dim(\mathcal{O}_i) = \dim(\mathcal{S}(\mathcal{O})),$$

as desired.

REMARK 2. In the compact case, the quantum version of functions on $\mathcal{M}(\mathcal{O}^{(1)}, \mathcal{O}^{(2)}, \mathcal{O}^{(3)})$ is the algebra of endomorphisms $End((V_{\lambda_1} \otimes V_{\lambda_2} \otimes V_{\lambda_3})^G)$ of the subspace of *G*-invariant vectors in the tensor product $V_{\lambda_1} \otimes V_{\lambda_2} \otimes V_{\lambda_3}$ with V_{λ_i} the representation corresponding to $\mathcal{O}^{(i)}$.

The quantum version of the algebra of functions on the fiber $\mathcal{P}(\mathcal{O}; \mathcal{O}^{(1)}, \ldots, \mathcal{O}^{(n)})$ is the algebra of endomorphisms of the vector space

 $\operatorname{Hom}_{G}(V_{\lambda_{1}}, V_{\lambda_{n}} \otimes V_{1}) \otimes \operatorname{Hom}_{G}(V_{\lambda_{2}}, V_{\lambda_{1}} \otimes V_{2}) \otimes \cdots \otimes \operatorname{Hom}_{G}(V_{\lambda_{n}}, V_{\lambda_{n-1}} \otimes V_{n}).$ Here the orbits \mathcal{O}_{i} correspond to V_{i} , and $\operatorname{Hom}_{G}(V_{\lambda_{i}}, V_{\lambda_{i-1}} \otimes V_{i})$ is the space of *G*-linear intertwiners $V_{\lambda_{i}} \to V_{\lambda_{i-1}} \otimes V_{i}$. For details see [**35**].

1.6. Constructing solutions by the projection method and angle variables. For \mathcal{H} a *G*-invariant function on \mathfrak{g}^* , write $\mathcal{H}^{(i)}$ for the G_n invariant function on $T^*(G^{\times n})$ defined by $\mathcal{H}^{(i)}(x,g) := \mathcal{H}(x_i)$. The Hamiltonian flow through $(x,g) \in \mathfrak{g}^{*\times n} \times G^{\times n}$ generated by $\mathcal{H}^{(i)}$ is

(25)
$$(x(t_i), g(t_i)) = (x_1, \dots, x_n, g_1, \dots, g_{i-1}, e^{\nabla \mathcal{H}(x_i)t_i}g_i, g_{i+1}, \dots, g_n)$$

where $\nabla \mathcal{H}(x) \in \mathfrak{g}$ is the gradient of \mathcal{H} at $x \in \mathfrak{g}^*$, i.e.,

$$y(\nabla \mathcal{H}(x)) = \frac{d}{dt}\mathcal{H}(x+ty)|_{t=0}$$

for all $y \in \mathfrak{g}^*$. The projection of such flow line to $T^*(G^{\times n})/G^{\times n}$ and further restricted to $\mathcal{S}(\mathcal{O}) \subset T^*(G^{\times n})/G_n$ is a flow line of the Hamiltonian vector field generated by the restriction of $\mathcal{H}^{(i)}$ to $\mathcal{S}(\mathcal{O})$.

Now let us construct angle variables, i.e., functions on $S(\mathcal{O})$ which evolve linearly with respect to the evolution (25) for each $i = 1, \ldots, n$. Write $\mathfrak{a}_+^* \subset \mathfrak{g}^*$ for the elements $x \in \mathfrak{g}^*$ which vanish on root spaces and satisfy $(x, \alpha) > 0$ for $\alpha \in R_+$, where (\cdot, \cdot) is the bilinear form on \mathfrak{g}^* induced by the Killing form. Write \mathfrak{g}'^* for the elements in \mathfrak{g}^* which are *G*-conjugate to an element in \mathfrak{a}_+^* , relative to the coadjoint action.

For $(x,g) \in \mathfrak{g}'^{*\times n} \times G^{\times n}$ define $s_i \in G$ by the property $Ad_{s_i}^*(x_i) \in \mathfrak{a}_+^*$. These elements are defined only up to $s_i \mapsto a_i s_i$ where $a_i \in H$. Gauge transformations $h \in G_n$ act by $(s_1, \ldots, s_n) \mapsto (s_1 h_1^{-1}, \ldots, s_n h_n^{-1})$.

Let $G_{\mathbb{C}}$ be a complexification of G, which we take to be connected. Let $H_{\mathbb{C}} \subset G_{\mathbb{C}}$ be the Cartan subgroup $Z_{G_{\mathbb{C}}}(\mathfrak{h})$, where \mathfrak{h} is the Cartan subalgebra $\mathfrak{a} \oplus i\mathfrak{a}$ of the Lie algebra $\mathfrak{g}_{\mathbb{C}} = \mathfrak{g} \oplus i\mathfrak{g}$ of $G_{\mathbb{C}}$. Then $A \subseteq H \subset H_{\mathbb{C}}$. We identify \mathfrak{g}^* with the real subspace of $\mathfrak{g}_{\mathbb{C}}^*$ consisting of the complex linear functionals that take real values on \mathfrak{g} . For finite dimensional $G_{\mathbb{C}}$ -representations V_1, \ldots, V_n choose vector $v_i \in V_i$ of $H_{\mathbb{C}}$ -weight λ_{i+1} and linear functionals $u_i^* \in V_i^*$ of $H_{\mathbb{C}}$ -weight $-\lambda_i$ (indices i taken modulo n).¹⁰ Define

(26)
$$f_{u,v}(x,g) = u_1^*(s_1g_1s_2^{-1}v_1)u_2^*(s_2g_2s_3^{-1}v_2)\dots u_n^*(s_ng_ns_1^{-1}v_n)$$

for $(x,g) \in (\mathfrak{g}'^* \times G)^{\times n}$. This expression is well defined (i.e., invariant with respect to transformations $s_i \to a_i s_i$ with $a_i \in H$), and invariant with respect to gauge transformations. Thus, it defines a function on the subset $(\mathfrak{g}'^{*\times n} \times G^{\times n})/G_n$ of $T^*(G^{\times n})/G_n$.

From the *G*-invariance of \mathcal{H} we have the identity

$$u_{i}^{*}(s_{i}e^{t_{i}\nabla\mathcal{H}(x_{i})}g_{i}s_{i+1}^{-1}v_{i}) = e^{t_{i}\lambda_{i}(\nabla\mathcal{H}(y_{i}))}u_{i}^{*}(s_{i}g_{i}s_{i+1}^{-1}v_{i})$$

where $y_i = Ad_{s_i}^*(x_i) \in \mathfrak{a}_+^*$, and consequently

(27)
$$f_{u,v}(x(t_i), g(t_i)) = e^{t_i \lambda_i (\nabla \mathcal{H}(y_i))} f_{u,v}(x, g).$$

Logarithms of these functions evolve linearly, and hence they produce angle variables for the Hamiltonians $\mathcal{H}^{(i)}$ on $S(\mathcal{O}) \cap (\mathfrak{g}^{\prime*\times n} \times G^{\times n})/G_n$.

2. Open spin Calogero-Moser chains

Recall from the introduction that G is a split real connected Lie group with finite center which admits a complexification, and $K \subset G$ is the subgroup of fixed points of a fixed Cartan involution Θ of G. Recall furthermore the root space decomposition $\mathfrak{g} = \mathfrak{a} \oplus \bigoplus_{\alpha>0} (\mathbb{R}e_{\alpha} \oplus \mathbb{R}e_{-\alpha})$ with the Cartan

¹⁰Recall that G acts on dual vectors as $(gu^*)(v) = u^*(g^{-1}v)$.

subalgebra $\mathfrak{a} \subset \mathfrak{g}$ and the root vectors $e_{\alpha} \in \mathfrak{g}_{\alpha}$ such that the infinitesimal Cartan involution θ acts as

$$\theta(h) = -h, \ \ \theta(e_{\alpha}) = -e_{-\alpha}$$

for $h \in \mathfrak{a}$ and $\alpha \in R$. We will furthermore normalise the root vectors in such a way that $(e_{\alpha}, e_{-\alpha}) = 1$, with (\cdot, \cdot) the Killing form of \mathfrak{g} .

To avoid cumbersome notations, we will not always indicate in notations that we are in the open case. This leads to an overlap of some of the notations with the ones for the periodic case. For instance, the moment maps, Poisson spaces and Poisson projections will be denoted in the same way.

2.1. The phase space as the Hamiltonian reduction. Consider for $n \ge 0$ the manifold $T^*(G^{\times n+1})$ with the standard symplectic structure. We trivialize the cotangent bundle $T^*(G^{\times n+1})$ by right translations:

(28)
$$T^*(G^{\times n+1}) \simeq (T^*G)^{\times n+1} \simeq \mathfrak{g}^{*\times n+1} \times G^{\times n+1}$$

We have a natural action of $K \times G^{\times n}$ on $G^{\times n+1}$ by left translations:

$$(k_{\ell}, h_1, \dots, h_n)_L(g_0, g_1, \dots, g_n) = (k_{\ell}g_0, h_1g_1, \dots, h_ng_n)$$

This action lifts to the following Hamiltonian action on $T^*G^{\times n+1}$,

$$(k_{\ell}, h_1, \dots, h_n)_L(x_0, \dots, x_n, g_0, g_1, \dots, g_n) = = (Ad_{k_{\ell}}^*(x_0), Ad_{h_1}^*(x_1), \dots, Ad_{h_n}^*(x_n), k_{\ell}g_0, h_1g_1, \dots, h_ng_n)$$

with the moment map

$$\mu_L(x,g) = (\pi(x_0), x_1, \dots, x_n)$$

where the projection $\pi : \mathfrak{g}^* \to \mathfrak{k}^*$ is the dual map dual to the embedding $\mathfrak{k} \hookrightarrow \mathfrak{g}$.

Similarly, the action of $G^{\times n} \times K$ on $G^{\times n+1}$ by right translations

$$(h_1, \ldots, h_n, k_r)_R(g_0, g_1, \ldots, g_n) = (g_0 h_1^{-1}, g_1 h_2^{-1}, \ldots, g_{n-1} h_n^{-1}, g_n k_r^{-1})$$

lifts to the following Hamiltonian action on $T^*G^{\times n+1}$,

$$(h_1, \dots, h_n, k_r)_R(x_0, \dots, x_n, g_0, g_1, \dots, g_n) = (x_0, x_1, \dots, x_n, g_0 h_1^{-1}, g_1 h_2^{-1}, \dots, g_{n-1} h_n^{-1}, g_n k_r^{-1}),$$

with the moment map

$$\mu_R(x,g) = (-Ad_{g_0^{-1}}^*(x_0), -Ad_{g_1^{-1}}^*(x_1), \dots, -Ad_{g_{n-1}^{-1}}^*(x_{n-1}), -\pi(Ad_{g_n^{-1}}^*(x_n))).$$

As a result, the group $G_{n,K} := K \times G^{\times n} \times K$ acts on $T^*(G^{\times n+1})$ as (29)

$$\begin{aligned} &(x_{\ell}, h_1, \dots, h_n, k_r)(x_0, \dots, x_n, g_0, \dots, g_n) = \\ &= (Ad_{k_{\ell}}^*(x_0), Ad_{h_1}^*(x_1), \dots, Ad_{h_n}^*(x_n), k_{\ell}g_0h_1^{-1}, h_1g_1h_2^{-1}, \dots, h_ng_nk_r^{-1}) \end{aligned}$$

with $k_{\ell}, k_r \in K$ and $h_i \in G$. This action is Hamiltonian with the moment map $\mu: T^*(G^{\times n}) \to \mathfrak{k}^* \times \mathfrak{g}^{* \times n} \times \mathfrak{k}^*$ given by

(30)

$$\mu((x_0, \dots, x_n, g_0, \dots, g_n) = (\mu_L(x, g), 0) + (0, \mu_R(x, g)) = \\
= (\pi(x_0), x_1 - Ad_{g_0^{-1}}^*(x_0), x_2 - Ad_{g_1^{-1}}^*(x_1), \dots, x_n - Ad_{g_{n-1}^{-1}}^*(x_{n-1}) \\
-\pi(Ad_{g_n^{-1}}^*(x_n))).$$

,

For n = 0 this is the $K \times K$ action $(k_{\ell}, k_r)(x, g) = (Ad_{k_{\ell}}^*(x), k_{\ell}gk_r^{-1})$ on T^*G , with the moment map $(x, g) \mapsto (\pi(x), -\pi(Ad_{g^{-1}}(x)))$. It is easy to check explicitly that this moment map intertwines the action of $G_{n,K}$ on $T^*(G^{\times n+1})$ given by (29) with its diagonal coadjoint action on $\mathfrak{k}^* \times \mathfrak{g}^{* \times (n-1)} \times \mathfrak{k}^*$.

Because the action of $G_{n,K}$ on $T^*(G^{\times n+1})$ is Hamiltonian, the space $T^*(G^{\times n+1})/G_{n,K}^{11}$ is Poisson with symplectic leaves being given by the Hamiltonian reduction with respect to the moment map (30). Let $\mathcal{O} = (\mathcal{O}_{\ell}^K \times \mathcal{O}_1 \times \ldots, \mathcal{O}_n \times \mathcal{O}_r^K)$ with $\mathcal{O}_i \subset \mathfrak{g}^*$ coadjoint *G*-orbits and $\mathcal{O}_{\ell}^K, \mathcal{O}_r^K \subset \mathfrak{t}^*$ coadjoint *K*-orbits, then the corresponding symplectic leaf in $T^*(G^{\times n+1})/G_{n,K}$ is

$$\begin{split} \mathcal{S}(\mathcal{O}) &= \mu^{-1}(\mathcal{O})/G_{n,K} \\ &= \left\{ (x_0, \dots, x_n, g_0, \dots, g_n) \in \mathfrak{g}^{* \times n+1} \times G^{\times n+1} \mid \pi(x_0) \in \mathcal{O}_{\ell}^K, \\ &- \pi(Ad_{g_n^{-1}}^{*-1}(x_n)) \in \mathcal{O}_r^K, \\ &x_1 - Ad_{g_0^{-1}}^{*-1}(x_0) \in \mathcal{O}_1, \dots, x_n - Ad_{g_{n-1}^{*-1}}^{*-1}(x_{n-1}) \in \mathcal{O}_n \right\}/G_{n,K} \end{split}$$

Each symplectic leaf $\S(\mathcal{O})$ is a stratified symplectic space and is the phase space for the corresponding open spin Calogero-Moser chain. We will describe the largest stratum of $\mathcal{S}(\mathcal{O})$ later.

2.2. The Hamiltonians of the open spin Calogero-Moser chain. After the trivialization (28) of $T^*(G^{\times n+1})$ by right translations we have a natural Poisson projection $T^*(G^{\times n+1}) \to \mathfrak{g}^{*\times n+1}$ to the first factor. It is $G_{n,K}$ -invariant with the following action of $G_{n,K}$ on $\mathfrak{g}^{*\times n+1}$

$$(k_{\ell}, h_1, \dots, h_n, k_r) : (x_0, x_1, \dots, x_n) \mapsto (Ad_{k_{\ell}}^*(x_0), Ad_{h_1}^*(x_1), \dots, Ad_{h_n}^*(x_n)).$$

This gives rise to the projection

$$p: T^*(G^{\times n+1})/G_{n,K} \to (\mathfrak{g}^*/G)^{\times n+1}$$

which is Poisson because it is the composition of natural Poisson projections (32)

$$T^*(G^{\times n+1})/G_{n,K} \to (\mathfrak{g}^{*\times n+1})/G_{n,K} = \mathfrak{g}^*/K \times (\mathfrak{g}^*/G)^{\times n} \to (\mathfrak{g}^*/G)^{\times n+1}$$

¹¹Recall that here and in everywhere else in this paper X/H means the GIT quotient for a Lie group H action on a manifold X.

Here the Poisson structure on the right is trivial (the Poisson tensor is zero). The last projection is a consequence of the embedding $K \hookrightarrow G$.

Restricting this projection to the symplectic leaf $\mathcal{S}(\mathcal{O})$ we have the Poisson projection

(33) $p_{\mathcal{O}}: \mathcal{S}(\mathcal{O}) \to \mathcal{B}(\mathcal{O}), \quad G_{n,K}(x_0, \ldots, x_n, g_0, \ldots, g_n) \mapsto (Gx_0, \ldots, Gx_n)$ where $\mathcal{B}(\mathcal{O}) \subset (\mathfrak{g}^*/G)^{\times n+1}$ is, by definition, the image of $\mathcal{S}(\mathcal{O})$. It can be described explicitly from the description of $\mathcal{S}(\mathcal{O})$ as

$$\mathcal{B}(\mathcal{O}) = \left\{ (\mathcal{O}^{(0)}, \dots, \mathcal{O}^{(n)}) \in (\mathfrak{g}^*/G)^{\times n+1} \mid \mathcal{O}_{\ell}^K \subseteq \pi(\mathcal{O}^{(0)}), \mathcal{O}_r^K \subseteq -\pi(\mathcal{O}^{(n)}), \\ \mathcal{O}_1 \subseteq \mathcal{O}^{(1)} - \mathcal{O}^{(0)}, \dots, \mathcal{O}_n \subseteq \mathcal{O}^{(n)} - \mathcal{O}^{(n-1)} \right\}.$$

Hamiltonians of the open spin Calogero-Moser system are pull back p^* of functions on $(\mathfrak{g}^*/G)^{\times n+1}$ restricted to $\mathcal{S}(\mathcal{O})$.

The subalgebra of Hamiltonians is a Poisson commuting subalgebra. Quadratic Hamiltonians are given by Casimir functions. We will compute the radial components of the quadratic Casimirs explicitly in section 2.5. Hamiltonians are constant on fibers of the projection $p_{\mathcal{O}}$.

2.3. The gauge fixing. Fix $i \in \{0, \ldots, n\}$. For $g = (g_0, \ldots, g_n) \in G^{\times n+1}$ and $k_{\ell}, k_r \in K$ define $h \in G^{\times n}$ (depending on i, g, k_{ℓ}, k_r) by

$$h_j = \begin{cases} k_\ell g_0 g_1 \cdots g_{j-1} & \text{if } j \le i, \\ k_r g_n^{-1} g_{n-1}^{-1} \cdots g_j^{-1} & \text{if } j > i. \end{cases}$$

Then $(k_{\ell}, h_1, \ldots, h_n, k_r)$ acts on $g \in G^{\times n+1}$ as

(34)

$$g \mapsto (1, \ldots, 1, k_{\ell} g_0 g_1 \ldots g_n k_r^{-1}, 1, \ldots, 1).$$

Here the nontrivial entry is at the position i. This gives an i-independent isomorphism

(35)
$$G^{\times n+1}/G_{n,K} \to K \setminus G/K, \qquad G_{n,K}(g_0, \dots, g_n) \mapsto Kg_0g_1 \dots g_nK.$$

For the cotangent bundles the gauge fixing gives an isomorphism

(36)
$$\phi_i: \left(\mathfrak{g}^{*\times n+1} \times G^{\times n+1}\right)/G_{n,K} \xrightarrow{\sim} K \setminus (\mathfrak{g}^{*\times n+1} \times G)/K$$

mapping the $G_{n,K}\text{-}\mathrm{orbit}$ through $(x,g)\in\mathfrak{g}^{*\times n+1}\times G^{\times n+1}$ to the double $K\text{-}\mathrm{coset}$ through

$$(x_0, Ad_{g_0}^*(x_1), \dots, Ad_{g_0\cdots g_{i-1}}^*(x_i), Ad_{g_n^{-1}\cdots g_{i+1}}^{*-1}(x_{i+1}), \dots, Ad_{g_n^{-1}}^{*-1}(x_n), g_0g_1\cdots g_n)$$

For example for i = n this expression is $(x_0, Ad_{g_0}^*(x_1), \ldots, Ad_{g_0\cdots g_{n-1}}^*(x_n), g_0g_1\cdots g_n)$. In (36) the double K-cosets in the codomain of ϕ_i are taken relative to the *i*-dependent $K \times K$ -action

$$(k_{\ell}, k_r)(x_0, \dots, x_n, g) = (Ad_{k_{\ell}}^*(x_0), \dots, Ad_{k_{\ell}}^*(x_i), Ad_{k_r}^*(x_{i+1}), \dots, Ad_{k_r}^*(x_n), k_{\ell}gk_r^{-1})$$

on $\mathfrak{g}^{* \times n+1} \times G$.

Now we can describe the symplectic leaf $\mathcal{S}(\mathcal{O})$ as a subvariety in $K \setminus (\mathfrak{g}^{* \times n+1} \times G)/K$ though the isomorphism φ_n as

(37)

$$\mathcal{S}(\mathcal{O}) = K \setminus \{ (y_0, y_1, \dots, y_n, g) \in \mathfrak{g}^{* \times n+1} \times G \mid \pi(y_0) \in \mathcal{O}_{\ell}^K, \\ -\pi (Ad_{g^{-1}}^*(y_n)) \in \mathcal{O}_r^K, \\ y_1 - y_0 \in \mathcal{O}_1, \dots, y_n - y_{n-1} \in \mathcal{O}_n \} / K.$$

Note that, as in the periodic case, $\mathcal{S}(\mathcal{O})$ is a symplectic stratified space. From now on we will focus mostly on the largest stratum $\mathcal{S}(\mathcal{O})_{reg}$.

2.4. The regular part of the symplectic leaf $\mathcal{S}(\mathcal{O})$. We use the gauge fixing isomorphism φ_n in the remainder of the text. We now use $K \setminus G/K \simeq A/W$ with $W = N_K(A)/M$ the Weyl group of G (see subsection §3 of the introduction) to describe the regular part of the symplectic leaf $\mathcal{S}(\mathcal{O})$ in radial coordinates.

Define the regular part of the phase space $\mathcal{S}(\mathcal{O})$ (see (37)) as

$$\mathcal{S}(\mathcal{O})_{reg} = \mathcal{S}(\mathcal{O}) \cap K \setminus (\mathfrak{g}^{* \times n+1} \times G_{reg}) / K$$

The regular part $\mathcal{S}(\mathcal{O})_{reg} \subset \mathcal{S}(\mathcal{O})$ is its largest stratum of the stratified symplectic space $\mathcal{S}(\mathcal{O})$.

We can then choose a representative of $K(y_0, \ldots, y_n, g)K \in \mathcal{S}(\mathcal{O})_{reg}$ with the *G*-component in A_{reg} by writing $g = k_{\ell}ak_r^{-1}$ and $y_i = Ad_{k_{\ell}}^*(x^{(i)})$ with $k_{\ell}, k_r \in K$ and $a \in A_{reg}$. It follows that

$$\mathcal{S}(\mathcal{O})_{reg} \simeq \left\{ (x^{(0)}, \dots, x^{(n)}, a) \in \mathfrak{g}^{* \times n+1} \times A_{reg} \mid \pi(x^{(0)}) \in \mathcal{O}_{\ell}^{K}, -\pi(Ad_{a^{-1}}^{*}(x^{(n)})) \in \mathcal{O}_{r}^{K}, x^{(1)} - x^{(0)} \in \mathcal{O}_{1}, \dots, x^{(n)} - x^{(n-1)} \in \mathcal{O}_{n} \right\} / N_{K}(A).$$

Here $N_K(A)$ acts diagonally on $\mathfrak{g}^{*\times n+1} \times A_{reg}$ via the coadjoint action on \mathfrak{g}^* and the conjugation action on A_{reg} .¹² We can now also divide out the action of $M = Z_K(A)$, to obtain the isomorphism

$$\begin{split} \mathcal{S}(\mathcal{O})_{reg} &\simeq \big\{ (M(x^{(0)}, \dots, x^{(n)}), a) \in \mathfrak{g}^{* \times n+1} / M \times A_{reg} \mid \pi(x^{(0)}) \in \mathcal{O}_{\ell}^{K}, \\ &- \pi(Ad^{*}_{a^{-1}}(x^{(n)})) \in \mathcal{O}_{r}^{K}, \\ &x^{(1)} - x^{(0)} \in \mathcal{O}_{1}, \dots, x^{(n)} - x^{(n-1)} \in \mathcal{O}_{n} \big\} / W, \end{split}$$

where M acts by the diagonal coadjoint action on $\mathfrak{g}^{*\times n+1}$ and W acts diagonally on the space $\mathfrak{g}^{*\times n+1}/M \times A_{reg}$ in the natural way.

Recall that we identified $\mathfrak{g} \simeq \mathfrak{g}^*$ and $\mathfrak{a} \simeq \mathfrak{a}^*$ via the Killing form, so that $x \in \mathfrak{g}^*$ corresponds to $x_0 + \sum_{\alpha} x_{\alpha} e_{\alpha}$ with x_0 the element in \mathfrak{a} corresponding

¹²We use here the fact that $k_{\ell}ak_r^{-1} = k'_{\ell}a'k'_r^{-1}$ for $k_{\ell}, k'_{\ell}, k_r, k'_r \in K$ and $a, a' \in A_{reg}$ is implying that $k_{\ell}^{-1}k'_{\ell} = k_r^{-1}k'_r \in N_K(A)$, cf., e.g., [20, §VII.3]. This essentially follows from the global Cartan decomposition of G.

to $x|_{\mathfrak{a}} \in \mathfrak{a}^*$ and $x_{\alpha} = x(e_{-\alpha})$. Denote by $x_0^{(k)}$, $x_{\alpha}^{(k)}$ the components of vectors $x \in \mathfrak{g}^{* \times n+1}$ from the k-th factor of \mathfrak{g}^{*n+1} , and $\mu_0^{(k)}$, $\mu_{\alpha}^{(k)}$ the components of $\mu \in \mathcal{O}_k$. For $y \in \mathfrak{k}^*$ we write $y_{[\alpha]} = y(e_{-\alpha} - e_{\alpha})$, so that $y_{[\alpha]} = -y_{[-\alpha]}$. Consider

$$T(\mathcal{O})_{reg} = \left\{ (x^{(0)}, \dots, x^{(n)}, a) \in \mathfrak{g}^{* \times n+1} \times A_{reg} \mid \pi(x^{(0)}) \in \mathcal{O}_{\ell}^{K}, \\ -\pi(Ad^{*}_{a^{-1}}(x^{(n)})) \in \mathcal{O}_{r}^{K}, \\ x^{(1)} - x^{(0)} \in \mathcal{O}_{1}, \dots, x^{(n)} - x^{(n-1)} \in \mathcal{O}_{n} \right\}.$$

Clearly $S(\mathcal{O})_{reg} = T(\mathcal{O})_{reg}/N_K(A).$

For $(x^{(0)}, \ldots, x^{(n)}, a) \in T(\mathcal{O})_{reg}$ write $\mu' = \pi(x^{(0)}) \in \mathcal{O}_{\ell}^{K}, \ \mu'' = -\pi(Ad^*_{a^{-1}}(x^{(n)})) \in \mathcal{O}_{r}^{K}$ and $\mu^{(i)} = x^{(i)} - x^{(i-1)} \in \mathcal{O}_{i}$ for $i = 1, \ldots, n$. The Cartan components of $x^{(k)}$ and their root coordinates then satisfy

(38)
$$\begin{aligned} x_{\alpha}^{(0)} - x_{-\alpha}^{(0)} &= \mu'_{[\alpha]}, \\ x_{\alpha}^{(i)} - x_{\alpha}^{(i-1)} &= \mu_{\alpha}^{(i)}, \\ x_{\alpha}^{(i)} - x_{\alpha}^{(i-1)} &= \mu_{\alpha}^{(i)}, \\ \end{aligned} \qquad \begin{array}{l} a_{\alpha} x_{-\alpha}^{(n)} - a_{\alpha}^{-1} x_{\alpha}^{(n)} &= \mu''_{[\alpha]}, \\ x_{0}^{(i)} - x_{0}^{(i-1)} &= \mu_{0}^{(i)} \\ \end{array}$$

for i = 1, ..., n.

It is easy to solve the equations for Cartan parts $x_0^{(i)}$ (0 < i < n) in terms of Cartan components of $x^{(0)}$, $x^{(n)}$ and $\mu^{(j)}$,

(39)
$$x_0^{(i)} = \mu_0^{(i)} + \dots + \mu_0^{(1)} + x_0^{(0)} = x_0^{(n)} - \mu_0^{(n)} - \dots - \mu_0^{(i+1)}$$

PROPOSITION 1. The following identities hold for $\alpha \in R$ and $k = 0, 1, \ldots, n$:

(40)
$$x_{\alpha}^{(k)} = K_{\alpha} + \sum_{l=1}^{k} \frac{a_{\alpha} \mu_{\alpha}^{(l)} - a_{\alpha} \mu_{-\alpha}^{(l)}}{a_{\alpha} - a_{\alpha}^{-1}} + \sum_{l=k+1}^{n} \frac{a_{\alpha}^{-1} \mu_{\alpha}^{(l)} - a_{\alpha} \mu_{-\alpha}^{(l)}}{a_{\alpha} - a_{\alpha}^{-1}}$$

where

(41)
$$K_{\alpha} = \frac{a_{\alpha}\mu'_{[\alpha]} + \mu''_{[\alpha]}}{a_{\alpha} - a_{\alpha}^{-1}}.$$

PROOF. Denote

(42)
$$\mu = \mu^{(1)} + \dots + \mu^{(n)}.$$

Note that $x^{(n)} - x^{(0)} = \mu$. Fix $\beta \in R_+$. For $x^{(1)}_{\pm\beta}$ and $x^{(n)}_{\pm\beta}$ the formula $x^{(n)} - x^{(0)} = \mu$ implies

$$x_{\beta}^{(n)} - x_{\beta}^{(0)} = \mu_{\beta}, \quad x_{-\beta}^{(n)} - x_{-\beta}^{(0)} = \mu_{-\beta}.$$

Combined with the first line of (38) we end up with four linear equations in $x_{\beta}^{(0)}, x_{-\beta}^{(0)}, x_{\beta}^{(n)}, x_{-\beta}^{(n)}, x_{-\beta}^{(n)}$ which, by the assumption that *a* is regular, are uniquely

solved by

(43)
$$x_{\alpha}^{(0)} = \frac{a_{\alpha}\mu'_{[\alpha]} + \mu''_{[\alpha]} + (a_{\alpha}^{-1}\mu_{\alpha} - a_{\alpha}\mu_{-\alpha})}{a_{\alpha} - a_{\alpha}^{-1}},$$
$$x_{\alpha}^{(n)} = \frac{a_{\alpha}\mu'_{[\alpha]} + \mu''_{[\alpha]} + (a_{\alpha}\mu_{\alpha} - a_{\alpha}\mu_{-\alpha})}{a_{\alpha} - a_{\alpha}^{-1}}$$

for $\alpha = \beta, -\beta$ (here we used that $a_{-\beta} = a_{\beta}^{-1}, \mu'_{[-\beta]} = -\mu'_{[\beta]}$ and $\mu''_{[-\beta]} = -\mu''_{[\beta]}$).

By the second line of (38) we then obtain

$$x_{\alpha}^{(k)} = \frac{a_{\alpha}\mu_{[\alpha]}' + \mu_{[\alpha]}'' + (a_{\alpha}^{-1}\mu_{\alpha} - a_{\alpha}\mu_{-\alpha}) + (a_{\alpha} - a_{\alpha}^{-1})(\mu_{\alpha}^{(1)} + \dots + \mu_{\alpha}^{(k)})}{a_{\alpha} - a_{\alpha}^{-1}}$$

for k = 0, 1, ..., n. Substituting (42) it is now easy to see that this is exactly what we wanted to prove.

The proposition and (39) give an isomorphism

(44)
$$S(\mathcal{O})_{reg} \simeq \left((\mathcal{O}_{\ell}^{K} \times \mathcal{O}_{1} \times \cdots \times \mathcal{O}_{n} \times \mathcal{O}_{r}^{K}) / M \times T^{*} A_{reg} \right) / W,$$

mapping $N_K(A)(x^{(0)}, \ldots, x^{(n)}, a)$ to the *W*-orbit of $(M(\mu', \mu^{(1)}, \ldots, \mu^{(n)}, \mu''), x_0^{(n)}, a)$, which preserves the natural symplectic structures. Here the finite discrete group $M = Z_K(A) \subset K$ acts diagonally via the coadjoint action, and $W = N_K(A)/M$ acts diagonally. The quantum version of this isomorphism is described in [35].

2.5. Quadratic Hamiltonians of open spin Calogero-Moser chain on the regular part of the phase space. In this section we compute the restriction of the Hamiltonian corresponding to the quadratic Casimir function on g^* ,

$$H_2^{(k)}(x,g) = \frac{1}{2}(x^{(k)}, x^{(k)}) = \frac{1}{2}(x_0^{(k)}, x_0^{(k)}) + \sum_{\alpha > 0} x_\alpha^{(k)} x_{-\alpha}^{(k)}$$

to the regular part of $\mathcal{S}(\mathcal{O})$ (see (31)) for $k = 0, \ldots, n$, where $(x, g) \in \mathfrak{g}^{* \times n+1} \times G^{\times n+1}$. Here (\cdot, \cdot) is the Killing form and $x_{\alpha}^{(i)}, x_{0}^{(i)}$ are the components of $x^{(i)}$ which were computed in the previous section on the regular part of the phase space.

We first consider the differences, which we will call the *boundary Knizhnik-Zamolodchikov-Bernard (bKZB) Hamiltonians*,

$$D_k = H_2^{(k)} - H_2^{(k-1)} \qquad (1 \le k \le n).$$

THEOREM 3. For the bKZB Hamiltonians we have the following formula:

(45)
$$D_k = (\mu_0^{(k)}, x_0^{(n)}) - \sum_{l=1}^{k-1} (r_{lk} + r_{lk}^{\theta_l}) + (\sum_{\alpha} K_{\alpha} \mu_{-\alpha}^{(k)} - \kappa_k) + \sum_{l=k+1}^n (r_{kl} - r_{kl}^{\theta_k}).$$

Here r_{kl} for $k \neq l$ is Felder's rescaled dynamical r-matrix

(46)
$$r_{kl} = -\frac{1}{2}(\mu_0^{(k)}, \mu_0^{(l)}) + \sum_{\alpha} \frac{\mu_{-\alpha}^{(k)} \mu_{\alpha}^{(l)}}{a_{\alpha}^2 - 1},$$

 θ_k is the transpose of the Chevalley involution θ acting on $\mu^{(k)}$,

$$\kappa_k = \frac{1}{2}(\mu_0^{(k)}, \mu_0^{(k)}) + \sum_{\alpha} \frac{(\mu_{\alpha}^{(k)})^2}{1 - a_{\alpha}^2}$$

is the core quadratic classical dynamical k-matrix and K_{α} is given by (41).

PROOF. The first step of the proof is the same as in the proof of Theorem 1, resulting in the expression

(47)
$$D_k = (\mu_0^{(k)}, x_0^{(k-1)} + \frac{1}{2}\mu_0^{(k)}) + \sum_{\alpha} x_{\alpha}^{(k-1)}\mu_{-\alpha}^{(k)} + \sum_{\alpha>0} \mu_{\alpha}^{(k)}\mu_{-\alpha}^{(k)}.$$

Now let us the formula (40) for $x_{\alpha}^{(k-1)}$,

(48)

$$\sum_{\alpha} x_{\alpha}^{(k-1)} \mu_{-\alpha}^{(k)} = \sum_{l=1}^{k-1} \sum_{\alpha} \frac{a_{\alpha} \mu_{\alpha}^{(l)} \mu_{-\alpha}^{(k)} - a_{\alpha} \mu_{-\alpha}^{(l)} \mu_{-\alpha}^{(k)}}{a_{\alpha} - a_{\alpha}^{-1}} + \sum_{\alpha} K_{\alpha} \mu_{-\alpha}^{(k)} + \sum_{\alpha} \frac{a_{\alpha}^{-1} \mu_{\alpha}^{(k)} \mu_{-\alpha}^{(k)} - a_{\alpha} \mu_{-\alpha}^{(k)} \mu_{-\alpha}^{(k)}}{a_{\alpha} - a_{\alpha}^{-1}} + \sum_{l=k+1}^{n} \sum_{\alpha} \frac{a_{\alpha}^{-1} \mu_{-\alpha}^{(k)} \mu_{\alpha}^{(l)} - a_{\alpha} \mu_{-\alpha}^{(k)} \mu_{-\alpha}^{(l)}}{a_{\alpha} - a_{\alpha}^{-1}}.$$

We express the different terms in the right hand side of (48) in terms of the dynamical *r*-matrix and *k*-matrix.

Note first that

$$r_{kl}^{\theta_k} = \frac{1}{2}(\mu_0^{(k)}, \mu_0^{(l)}) - \sum_{\alpha} \frac{\mu_{\alpha}^{(k)} \mu_{\alpha}^{(l)}}{a_{\alpha}^2 - 1} = r_{lk}^{\theta_l}$$

Then the terms in the right hand side of (48) with l strictly smaller than k can be rewritten as

$$\sum_{\alpha} \frac{a_{\alpha} \mu_{\alpha}^{(l)} \mu_{-\alpha}^{(k)} - a_{\alpha} \mu_{-\alpha}^{(l)} \mu_{-\alpha}^{(k)}}{a_{\alpha} - a_{\alpha}^{-1}} = -(r_{lk} + r_{lk}^{\theta_l})$$

while the terms in the right hand side of (48) with l strictly larger than k reduce to

$$\sum_{\alpha} \frac{a_{\alpha}^{-1} \mu_{-\alpha}^{(k)} \mu_{\alpha}^{(l)} - a_{\alpha} \mu_{-\alpha}^{(k)} \mu_{-\alpha}^{(l)}}{a_{\alpha} - a_{\alpha}^{-1}} = (\mu_0^{(k)}, \mu_0^{(l)}) + (r_{kl} - r_{kl}^{\theta_k}).$$

Finally, for the middle term in (48) a direct computation shows that

$$\sum_{\alpha} \frac{a_{\alpha}^{-1} \mu_{\alpha}^{(k)} \mu_{-\alpha}^{(k)} - a_{\alpha} \mu_{-\alpha}^{(k)} \mu_{-\alpha}^{(k)}}{a_{\alpha} - a_{\alpha}^{-1}} = \frac{1}{2} (\mu_0^{(k)}, \mu_0^{(k)}) - \sum_{\alpha > 0} \mu_{\alpha}^{(k)} \mu_{-\alpha}^{(k)} - \kappa_k.$$

Substitute these formulas in (48), then the resulting formula (47) for D_k becomes

$$D_{k} = (\mu_{0}^{(k)}, x_{0}^{(k-1)} + \mu_{0}^{(k)} + \mu_{0}^{(k+1)} + \dots + \mu_{0}^{(n)}) - \sum_{l=1}^{k-1} (r_{lk} + r_{lk}^{\theta_{l}}) + (\sum_{\alpha} K_{\alpha} \mu_{-\alpha}^{(k)} - \kappa_{k}) + \sum_{l=k+1}^{n} (r_{kl} - r_{kl}^{\theta_{k}}).$$

By (39) this reduces to the formula (45).

The quantum versions of the boundary KZB Hamiltonians in the present context were obtained in $[35, \S6]$. It was extended to the case of non-split real semisimple Lie groups G in [33].

For the Hamiltonian $H_2^{(n)}$ we obtain by (43) the expression

$$H_2^{(n)} = \frac{1}{2}(p,p) + \sum_{\alpha>0} \frac{(a_\alpha \mu'_{[\alpha]} + \mu''_{[\alpha]} + a_\alpha (\mu_\alpha - \mu_{-\alpha}))(a_\alpha^{-1} \mu'_{[\alpha]} + \mu''_{[\alpha]} + a_\alpha^{-1} (\mu_\alpha - \mu_{-\alpha}))}{(a_\alpha - a_{-\alpha})^2}$$

on $\mathcal{S}(\mathcal{O})_{reg}$, where $\mu = \mu^{(1)} + \cdots + \mu^{(n)}$. Here we use notation $p = x_0^{(n)}$ for the cotangent vectors to A_{reg} in formula (44). Note that the potential term only depends on the restrictions $\pi(\mu^{(i)})$ of $\mu^{(i)} \in \mathfrak{g}^*$ to \mathfrak{k} , since $\mu_{\alpha} - \mu_{-\alpha} = -(\pi(\mu))_{[\alpha]}$. The radial component of the quantum quadratic Hamiltonian in the current open context was obtained in [35, §6].

2.6. The superintegrability of the open spin CM chain. In this section we will prove that Poisson commutative subalgebra of Hamiltonians constructed in section 2.2 defines a superintegrable system. Fix $\mathcal{O} = (\mathcal{O}_{\ell}^{K}, \mathcal{O}_{1}, \ldots, \mathcal{O}_{n}, \mathcal{O}_{r}^{K}) \in \mathfrak{k}^{*}/K \times (\mathfrak{g}^{*}/G)^{\times n} \times \mathfrak{k}^{*}/K$. We will construct Poisson projections

(49)
$$S(\mathcal{O}) \xrightarrow{p_{1,\mathcal{O}}} \mathcal{P}(\mathcal{O}) \xrightarrow{p_{2,\mathcal{O}}} \mathcal{B}(\mathcal{O})$$

such that $p_{\mathcal{O}} = p_{2,\mathcal{O}} \circ p_{1,\mathcal{O}}$ (see (33)), satisfying the desired properties.

Let $(\mathfrak{k}^* \times \mathfrak{g}^{* \times n}) \times_{(\mathfrak{g}^*/G)^{\times n+1}} (\mathfrak{g}^{* \times n} \times \mathfrak{k}^*)$ be the subset of $(\mathfrak{k}^* \times \mathfrak{g}^{* \times n}) \times (\mathfrak{g}^{* \times n} \times \mathfrak{k}^*)$ consisting of elements $(z_{\ell}, x_1, \ldots, x_n, y_1, \ldots, y_n, z_r)$ satisfying

$$z_{\ell} \in -\pi(Gy_1), \quad x_i \in -Gy_{i+1} \ (1 \le i < n), \quad z_r \in -\pi(Gx_n).$$

The gauge group $G_{n,K}$ acts on $(\mathfrak{k}^* \times \mathfrak{g}^{*\times n}) \times_{(\mathfrak{g}^*/G)^{\times n+1}} (\mathfrak{g}^{*\times n} \times \mathfrak{k}^*)$ by (50)

$$\begin{aligned} &(k_{\ell}, h_1, \dots, h_n, k_r)(z_{\ell}, x_1, \dots, x_n, y_1, \dots, y_n, z_r) = \\ &= (Ad_{k_{\ell}}^* z_{\ell}, Ad_{h_1}^* x_1, \dots, Ad_{h_n}^* x_n, Ad_{h_1}^* y_1, \dots, Ad_{h_n}^* y_n, Ad_{k_r}^* z_r). \end{aligned}$$

Consider the resulting Poisson space

$$\mathcal{P} = \left(\left(\mathfrak{k}^* \times \mathfrak{g}^{* \times n} \right) \times_{\left(\mathfrak{g}^*/G \right)^{\times n+1}} \left(\mathfrak{g}^{* \times n} \times \mathfrak{k}^* \right) \right) / G_{n,K}$$

and define the Poisson map

$$p_1: T^*(G^{\times n+1})/G_{n,K} \to \mathcal{P}$$

by

$$p_1(G_{n,K}(x,g)) = G_{n,K}(\mu_L(x,g),\mu_R(x,g))$$

= $G_{n,K}(\pi(x_0),x_1,\ldots,x_n,-Ad_{g_0^{-1}}^*(x_0),\ldots,$
 $-Ad_{g_{n-1}^{-1}}^*(x_{n-1}),-\pi(Ad_{g_n^{-1}}^*(x_n))).$

Here $(x,g) = (x_0, \ldots, x_n, g_0, \ldots, g_n) \in \mathfrak{g}^{* \times n+1} \times G^{\times n+1} \simeq T^*(G^{\times n+1})$. Define the Poisson projection

$$p_2: \mathcal{P} \to (\mathfrak{g}^*/G)^{\times n+1}$$

by

$$p_2(G_{n,K}(z_\ell, x_1, \dots, x_n, y_1, \dots, y_n, z_r)) = (-Gy_1, Gx_1, \dots, Gx_n),$$

with the trivial Poisson structure on the target space.

The restriction of the Poisson projection p_1 to the symplectic leaf $S(\mathcal{O}) \subset T^*(G^{\times n+1})/G_{n,K}$ (see (31)) gives the Poisson projection

$$p_{1,\mathcal{O}}: S(\mathcal{O}) \to \mathcal{P}(\mathcal{O})$$

where

$$\mathcal{P}(\mathcal{O}) = (\mu_L \times \mu_R)(\mu^{-1}(\mathcal{O}))/G_{n,K},$$

or, more explicitly,

(51)

$$\mathcal{P}(\mathcal{O}) = \left\{ (z_{\ell}, x_1, \dots, x_n, y_1, \dots, y_n, z_r) \in (\mathfrak{k}^* \times \mathfrak{g}^{* \times n}) \times_{(\mathfrak{g}^*/G)^{\times n+1}} (\mathfrak{g}^{* \times n} \times \mathfrak{k}^*) \mid z_{\ell} \in \mathcal{O}_{\ell}^K, x_1 + y_1 \in \mathcal{O}_1, \dots, x_n + y_n \in \mathcal{O}_n, z_r \in \mathcal{O}_r^K \right\} / G_{n,K}.$$

The generic fibers of this mapping are isotropic submanifolds in $\mathcal{S}(\mathcal{O})$.

The restriction of the Poisson projection p_2 to $\mathcal{P}(\mathcal{O})$ gives a surjective Poisson projection

$$p_{2,\mathcal{O}}: \mathcal{P}(\mathcal{O}) \to \mathcal{B}(\mathcal{O}),$$

with $\mathcal{B}(\mathcal{O})$ given by (34). Clearly, the composition of $p_{2,\mathcal{O}} \circ p_{1,\mathcal{O}} : \mathcal{S}(\mathcal{O}) \to \mathcal{B}(\mathcal{O})$ is the projection $p_{\mathcal{O}}$ as given by (33).

Now let us describe fibers of $p_{2,\mathcal{O}}$ are symplectic leaves of $\mathcal{P}(\mathcal{O})$.

LEMMA 3. We have he following symplectomorphysm

(52)
$$p_{2,\mathcal{O}}^{-1}((\mathcal{O}^{(0)},\ldots,\mathcal{O}^{(n)})) \xrightarrow{\sim} \mathcal{M}(-\mathcal{O}^{(0)},\mathcal{O}_{\ell}^{K}) \times \prod_{i=1}^{n} \mathcal{M}(-\mathcal{O}^{(i)},\mathcal{O}^{(i-1)},\mathcal{O}_{i}) \times \mathcal{M}(\mathcal{O}^{(n)},\mathcal{O}_{r}^{K})$$

where symplectic spaces $\mathcal{M}(-\mathcal{O}^{(i)}, \mathcal{O}^{(i-1)}, \mathcal{O}_i)$ are defined in (22) and

$$\mathcal{M}(\mathcal{O}', \mathcal{O}^K) = \{(x, z) \in \mathcal{O}' \times \mathcal{O}^K \mid \pi(x) + z = 0\}/K$$

Here $\mathcal{O} \subset \mathfrak{g}^*$ is a G-coadjoint orbit and $\mathcal{O}^K \subset \mathfrak{k}^*$ is a K-coadjoint orbit. It has a natural symplectic structure because it is the Hamiltonian reduction of $\mathcal{O} \times \mathcal{O}^K$ with respect to the Hamiltonian diagonal action of K.

PROOF. Let $(\mathcal{O}^{(0)}, \ldots, \mathcal{O}^{(n)}) \in \mathcal{B}(\mathcal{O})$. By a direct computation, the fiber $p_{2,\mathcal{O}}^{-1}((\mathcal{O}^{(0)}, \ldots, \mathcal{O}^{(n)}))$ consists of the $G_{n,K}$ -orbits in \mathcal{P} with representatives

$$(z_{\ell}, x_1, \dots, x_n, y_1, \dots, y_n, z_r) \in (\mathfrak{k}^* \times \mathfrak{g}^{* \times n}) \times (\mathfrak{g}^{* \times n} \times \mathfrak{k}^*)$$

satisfying the following conditions

$$z_{\ell} \in \mathcal{O}_{\ell}^{K} \cap \pi(\mathcal{O}^{(0)}), \qquad x_{i} + y_{i} \in \mathcal{O}_{i} \quad (1 \le i \le n), \qquad z_{r} \in \mathcal{O}_{r}^{K} \cap -\pi(\mathcal{O}^{(n)}),$$
$$-y_{1} \in \mathcal{O}^{(0)}, \qquad x_{i} \in \mathcal{O}^{(i)}, -y_{i+1}\mathcal{O}^{(i)} \in (1 \le i \le n-1), \qquad x_{n} \in \mathcal{O}^{(n)}.$$

Using this explicit description of the fiber, we can write it as a direct product of symplectic spaces. The isomorphism (23) for symplectic spaces $\mathcal{M}(\mathcal{O}^{(1)}, \mathcal{O}^{(2)}, \mathcal{O}^{(3)})$ defined by (22) gives factors $\mathcal{M}(-\mathcal{O}^{(i)}, \mathcal{O}^{(i-1)}, \mathcal{O}_i)$. The space

$$\mathcal{M}(\mathcal{O}', \mathcal{O}^K) = \{(x, z) \in \mathcal{O}' \times \mathcal{O}^K \mid \pi(x) + z = 0\}/K$$

with K acting by the diagonal coadjoint action is symplectic because, see above. The isomorphism

$$\mathcal{M}(\mathcal{O}', \mathcal{O}^K) \xrightarrow{\sim} (\mathcal{O}' \cap \pi^{-1}(\mathcal{O}^K))/K,$$

completes the proof. The isomorphism maps $G_{n,K}(z_{\ell}, x_1, \ldots, x_n, y_1, \ldots, y_n, z_r) \in p_{2,\mathcal{O}}^{-1}((\mathcal{O}^{(0)}, \ldots, \mathcal{O}^{(n)}))$ to

$$(K(\widetilde{y}_1, z_\ell), G(-x_1, -y_1, x_1 + y_1), \dots, G(-x_n, -y_n, x_n + y_n), K(\widetilde{x}_n, z_r)),$$

with $\widetilde{y}_1 \in Gy_1 = -\mathcal{O}^{(0)}$ such that $-\pi(\widetilde{y}_1) = z_\ell$ and $\widetilde{x}_n \in Gx_n = \mathcal{O}^{(n)}$ such that $-\pi(\widetilde{x}_n) = z_r$.

Note that for generic \mathcal{O}' , the symplectic space $\mathcal{M}(\mathcal{O}', \mathcal{O}^K)$ is of dimension dim (\mathcal{O}^K) .

REMARK 3. In the compact case, the algebra of function on the fiber of $p_{2,\mathcal{O}}$ has the algebra of endomorphisms of the vector space

$$\operatorname{Hom}_{K}(V_{\lambda_{0}}, U_{\nu_{\ell}}) \otimes \operatorname{Hom}_{G}(V_{\lambda_{1}}, V_{\lambda_{0}} \otimes V_{\mu_{1}}) \otimes \cdots \otimes \operatorname{Hom}_{G}(V_{\lambda_{n}}, V_{\lambda_{n-1}} \otimes V_{\mu_{n}}) \\ \otimes \operatorname{Hom}_{K}(U_{\nu_{\tau}}, V_{\lambda_{n}})$$

as natural quantization, where the finite dimensional *G*-representation V_{λ_i} (resp. V_{μ_i}) corresponds to $\mathcal{O}_{\ell}^{(i)}$ (resp. \mathcal{O}_i) and the *K*-representation $U_{\nu_{\ell}}$ (resp. U_{ν_r}) corresponds to \mathcal{O}_{ℓ}^K (resp. \mathcal{O}_r^K), compare with Remark 2 in the cyclic case. For details see [35] (which treats the noncompact case) and [31].

LEMMA 4. Dimensions of spaces $\mathcal{B}(\mathcal{O})$ and $\mathcal{P}(\mathcal{O})$ are

$$\dim(\mathcal{B}(\mathcal{O})) = (n+1)r, \quad \dim(\mathcal{P}(\mathcal{O})) = \dim(\mathcal{O}) - 2nr$$

where we define dim(\mathcal{O}) as dim(\mathcal{O}_{ℓ}^{K}) + $\sum_{i=1}^{n} \dim(\mathcal{O}_{i})$ + dim(\mathcal{O}_{r}^{K}).

PROOF. The proof of the dimension formula for $\mathcal{B}(\mathcal{O})$ is completely similar to the periodic case. It is enough to consider large orbits. For the dimension of $\mathcal{P}(\mathcal{O})$ we have:

$$\dim(\mathcal{P}(\mathcal{O})) = \dim(\mathcal{B}(\mathcal{O})) + \dim(p_{2,\mathcal{O}}^{-1}(\mathcal{O}^{(0)},\ldots,\mathcal{O}^{(n)}))$$

and by (52) the dimension of $p_{2,\mathcal{O}}^{-1}((\mathcal{O}^{(0)},\ldots,\mathcal{O}^{(n)}))$ is equal to

$$\dim(\mathcal{M}(-\mathcal{O}^{(0)},\mathcal{O}_{\ell}^{K})) + \sum_{i=1}^{n} \dim(\mathcal{M}(-\mathcal{O}^{(i)},\mathcal{O}^{(i-1)},\mathcal{O}_{i})) + \dim(\mathcal{M}(\mathcal{O}^{(n)},\mathcal{O}_{r}^{K})) =$$
$$= \dim(\mathcal{O}_{\ell}^{K}) + \sum_{i=1}^{n} (\dim(\mathcal{O}_{i}) - 2r) + \dim(\mathcal{O}_{r}^{K}) = \dim(\mathcal{O}) - 2nr.$$

This finishes the proof.

We now have the following main result of this section.

THEOREM 4. The Hamiltonian system generated by any Hamiltonian for the open spin CM chain described in section 2.2 is superintegrable with the superintegrable structure described by the surjective Poisson maps

$$\mathcal{S}(\mathcal{O}) \xrightarrow{p_{1,\mathcal{O}}} \mathcal{P}(\mathcal{O}) \xrightarrow{p_{2,\mathcal{O}}} \mathcal{B}(\mathcal{O})$$

as introduced earlier in this section.

Recall that $\mathcal{O}_i \neq \{0\}$ for all $i = 0, 1, \dots, n$.

PROOF. We already verified most of the conditions. What remains to show is the matching of dimensions,

(53)
$$\dim(\mathcal{S}(\mathcal{O})) = \dim(\mathcal{P}(\mathcal{O})) + \dim(\mathcal{B}(\mathcal{O})).$$

For the collection $\mathcal{O} = (\mathcal{O}_{\ell}^{K}, \mathcal{O}_{1}, \dots, \mathcal{O}_{n}, \mathcal{O}_{r}^{K})$ of coadjoint orbits we write $\dim(\mathcal{O})$ for the sum of the dimensions of the coadjoint orbits.

For the dimension of the symplectic leaf $\mathcal{S}(\mathcal{O})$ we have, using (44),

$$\dim(\mathcal{S}(\mathcal{O})) = 2r + \dim(\mathcal{O}),$$

with r the rank of \mathfrak{g} . For $\mathcal{P}(\mathcal{O})$ we obtain for generic $(\mathcal{O}^{(0)}, \ldots, \mathcal{O}^{(n)}) \in \mathcal{B}(\mathcal{O})$. Because

$$\dim(\mathcal{B}(\mathcal{O})) = (n+1)r,$$

we have

$$\dim(\mathcal{P}(\mathcal{O})) + \dim(\mathcal{B}(\mathcal{O})) = \dim(\mathcal{O}) - 2nr + 2\dim(\mathcal{B}(\mathcal{O}))$$
$$= \dim(\mathcal{O}) + 2r$$
$$= \dim(\mathcal{S}(\mathcal{O})),$$

as desired.

2.7. Constructing solutions by the projection method and angle variables. Let \mathcal{H} be a *G*-invariant function on \mathfrak{g}^* and $\mathcal{H}^{(i)}$ for $i = 0, \ldots, n$ the associated G_n -invariant function $(x, g) \mapsto \mathcal{H}(x_i)$ on $T^*(G^{\times n+1})$ (cf. section 1.6). The Hamiltonian flow generated by $\mathcal{H}^{(i)}$ on $T^*(G^{\times n+1}) \simeq \mathfrak{g}^{*\times n+1} \times G^{\times n+1}$ was already described in section 1.6. The flow line passing through (x, g) at t = 0 is

(54)
$$(x(t_i), g(t_i)) = (x_0, \dots, x_n, g_0, \dots, g_{i-1}, e^{\nabla \mathcal{H}(x_i)t_i}g_i, g_{i+1}, \dots, g_n).$$

The corresponding Hamiltonian flow on the symplectic leaf

$$\mathcal{S}(\mathcal{O}) \subset T^*(G^{\times n+1})/G_{n,K}$$

is obtained by projecting the flow (54) to $T^*(G^{\times n+1})/G_{n,K}$ and restricting it to $\mathcal{S}(\mathcal{O})$. Thus, we reformulated the problem of solving nonlinear differential equations of motion for open spin Calogero-Moser chains to a problem of linear algebra. This is a version of the original projection method which goes back to earlier papers on Calogero-Moser type systems [28].

Now let us describe angle variables for this integrable dynamics. We use the notations from section 1.6. Fix $(x, g) \in \mathfrak{g}'^{* \times n+1} \times G^{\times n+1}$. For $i = 1, \ldots, n$ define elements $s_i \in G$ by the condition $Ad_{s_i}^*(x_i) \in \mathfrak{a}_+^*$. As in the periodic case (see section 1.6) it is defines up to $s_i \mapsto a_i s_i$ with $a_i \in H \subset G$, where $H \subset G$ is the Cartan subgroup containing A. Define $s_0 \in K$ such that $Ad_{s_0}^*(x_0)|_{\mathfrak{p}} \in \mathfrak{a}_+^*$ (where we view \mathfrak{a}_+^* now as subset of \mathfrak{p}^* in the natural manner). The element s_0 is defined up to $s_0 \mapsto ms_0$ with $m \in M = Z_K(A)$. Similarly, we define $s_{n+1} \in K$ such that $Ad_{s_{n+1}}^*(x_n)|_{\mathfrak{p}} \in \mathfrak{a}_+^*$.

Choose finite dimensional representations V_0, V_1, \ldots, V_n of $G_{\mathbb{C}}, H_{\mathbb{C}}$ -weight vectors $v_i \in V_i$ of weight λ_{i+1} for $0 \leq i < n$ and $H_{\mathbb{C}}$ -weight vectors $u_j^* \in V_j^*$ of weight $-\lambda_j$ for $0 \leq j \leq n$. Finally, we choose *M*-invariant vectors $u_0^* \in V_0^*$ and $v_n \in V_n$ (i.e., $mu_0^* = u_0^*$ and $mv_n = v_n$ for all $m \in M$). Define

(55)
$$f_{u,v}(x,g) = u_0^*(s_0g_0s_1^{-1}v_0)u_1^*(s_1g_1s_2^{-1}v_1)\cdots u_{n-1}^*(s_{n-1}g_{n-1}s_n^{-1}v_{n-1})u_n^*(s_ng_ns_{n+1}^{-1}v_n).$$

It is an easy check that $f_{u,v}(x,g)$ is a well defined $G_{n,K}$ -invariant function on $\mathfrak{g}'^{*\times n+1} \times G^{\times n+1}$.

Similarly as in the periodic case (see section 1.6) we then have for $i = 1, \ldots, n$,

(56)
$$f_{u,v}(x(t_i), g(t_i)) = e^{t_i \lambda_i (\nabla \mathcal{H}(y_i))} f_{u,v}(x, g)$$

with $y_i = Ad_{s_i}^*(x_i) \in \mathfrak{a}_+^*$. Logarithms of these functions thus evolve linearly, and hence give rise to angle variables for the Hamiltonians $\mathcal{H}^{(i)}$ on $\mathcal{S}(\mathcal{O}) \cap (\mathfrak{g}'^{*\times n+1} \times G^{\times n+1})/G_{n,K}$.

For i = 0 we need to restrict further to $(x, g) \in \mathfrak{g}^{\prime * \times n+1} \times G^{\times n+1}$ with $x_0 \in \mathfrak{p}$, and assume that $u_0^* \in V_0^*$ is not only *M*-invariant but also a $H_{\mathbb{C}}$ -weight vector, say of weight $-\lambda_0$. In this case $\operatorname{Ad}_{s_0}^*(x_0) = y_0 \in \mathfrak{a}_+^*$ and hence

$$u_0^*(s_0e^{t_0\nabla\mathcal{H}(x_0)}g_0s_1^{-1}v_0) = e^{t_0\lambda_0(\mathcal{H}(y_0))}u_0^*(s_0g_0s_1^{-1}v_0).$$

As a consequence (56) then also holds true for i = 0, and the logarithm of $f_{u,v}(x,g)$ becomes a linear functions of time t_0 .

3. A Liouville integrable example of a periodic spin Calogero-Moser example for orbits of rank 1

3.1. Let us briefly discuss a particular case of periodic spin CM chain corresponding to $G = SL_N(\mathbb{R})$ with rank one orbits \mathcal{O}_k . This case is related to the original paper [16] where spin CM systems were first introduced.

Take $\mathfrak{a} \subset \mathfrak{sl}_N$ the Cartan subalgebra consisting of diagonal matrices, and denote the roots by $\{\epsilon_i - \epsilon_j\}_{i \neq j} \subset \mathfrak{a}^*$ with $\epsilon_i \in \mathfrak{a}^*$ the linear functional picking out the i^{th} diagonal entry. We identify \mathfrak{sl}_N with its dual via the Killing form $(x, y) = \operatorname{Tr}(xy)$. Then for $p \in \mathfrak{a}^* \simeq \mathfrak{a}$ we have $(p, p) = \sum_{i=1}^N p_i^2$, with p_i the i^{th} diagonal entry of the diagonal matrix p. For $y \in \mathfrak{sl}_N^* \simeq \mathfrak{sl}_N$ and $i \neq j$ we have $y_{\epsilon_i - \epsilon_j} = y_{ij}$, with y_{ij} the $(i, j)^{th}$ entry of the matrix y.

For $\xi \in \mathbb{R}$ set

$$\mathcal{O}^{(\xi)} = \left\{ z - \frac{\xi}{N} \mathrm{id}_N \mid z \text{ is a rank one } N \times N \text{ matrix with } \mathrm{Tr}(z) = \xi \right\}.$$

Then $\mathcal{O}^{(\xi)}$ is a coadjoint orbit in $\mathfrak{sl}_N \simeq \mathfrak{sl}_N^*$ of dimension 2(N-1).

Viewing elements in \mathbb{R}^N as column vectors, we have a natural mapping

(57)
$$\{(a,b) \in \mathbb{R}^N \times \mathbb{R}^N \mid a^t b = \sum_{i=1}^N a_i b_i = \xi \} / \mathbb{R}^{\times} \xrightarrow{\sim} \mathcal{O}^{(\xi)},$$
$$\mathbb{R}^{\times}(a,b) \mapsto \mu = b a^t - \frac{\xi}{N} \mathrm{id}_N$$

where $\lambda \in \mathbb{R}^{\times}$ acts by $(a, b) \mapsto (\lambda a, \lambda^{-1}b)$ and a^{t} is the transpose of $a \in \mathbb{R}^{N}$, so that $\mu_{ij} = a_{i}b_{j} - \delta_{ij}\frac{\zeta}{N}$. Because of the rank one condition, this is an isomorphism. It is easy to check that this is symplomorphism, with the Poisson brackets of the coordinate functions a_{i} and b_{j} of $(a, b) \in \mathbb{R}^{N} \times \mathbb{R}^{N}$ given by

$$\{b_i, a_j\} = \delta_{ij}, \ \{a_i, a_j\} = 0 = \{b_i, b_j\} = 0.$$

The value of the quadratic Casimir function $(y, y) = \sum_{i,j=1}^{N} y_{ij} y_{ji}$ being restricted to $\mathcal{O}^{(\xi)}$ can easily computed using (57):

$$\sum_{i,j=1}^{N} \mu_{ij} \mu_{ji} = \xi^2 \left(1 - \frac{1}{N} \right), \qquad \mu \in \mathcal{O}^{(\xi)}.$$

Note that the action of \mathbb{R}^{\times} in (57) is Hamiltonian and that $\mathcal{O}^{(\xi)}$ can be regarded as the Hamiltonian reduction of $T^*\mathbb{R}^N$ with coordinates a_i on \mathbb{R}^N and b_i on cotangent spaces with respect the action and the moment map

$$(a,b) \mapsto H_{\xi}(a,b) = \sum_{i=1}^{N} a_i b_i - \xi$$

3.2. Now consider *n*-spin Calogero-Moser system for $G = SL_N$ with the phase space $\mathcal{S}(\mathcal{O})$ corresponding to $\mathcal{O} = \mathcal{O}_{\xi_1} \times \cdots \times \mathcal{O}_{\xi_n}$. The regular part of this phase space is an open dense subset $\mathcal{S}(\mathcal{O})_{reg} \subset \mathcal{S}(\mathcal{O})$ that we described earlier by an isomorphism

$$\mathcal{S}(\mathcal{O})_{reg} \simeq (T^* A_{N,reg} \times \mathcal{O}//A)/W$$

where $\mathcal{O}//A$ is the Hamiltonian reduction of \mathcal{O} with respect to the diagonal action of the Cartan subgroup $A_N \subset SL_N$ and $A_{N,reg} \subset A_N$ are diagonal matrices with positive pairwise distinct entries.

Quadratic Hamiltonians were described in section 1.4. The quadratic *n*-th Hamiltonian $H_2^{(n)}$ has a particularly simple form

(58)
$$H_2 = \frac{1}{2} \sum_{i=1}^{N} p_i^2 - \sum_{i < j} \frac{\mu_{ij} \mu_{ji}}{2 \mathrm{sh}^2 (q_i - q_j)}$$

where $q_i = \epsilon_i(\log(a))$ are natural logarithmic coordinates on the Cartan subgroup $A_N \in SL_N(\mathbb{R})$, p_i are corresponding coordinates on the cotagent spaces in $T^*A_{N,reg}$ and

$$\mu_{ij} = \sum_{k=1}^{n} \mu_{ij}^{(k)}.$$

Here for each k = 1, ..., n, $\mu_{ij}^{(k)}$ is the restriction of natural coordinates on sl_N^* to \mathcal{O}_{ξ_k} regarded as a function of the k-th factor of $\mathcal{O}_{\xi_1} \times \cdots \times \mathcal{O}_{\xi_n}$.

Now let us describe the "spin" part of the phase space as the Hamiltonian reduction with respect to the action of $A_N \times A_n$ where $A_N \subset SL_N$ and $A_n \subset SL_n$ are Cartan subgroups (groups of unimodular diagonal matrices). For n = 1 we already did this in section 3.1.

As it is explained above we can write $\mu_{ij}^{(k)} = b_i^{(k)} a_j^{(k)} - \frac{\xi_k}{N} \delta_{ij}$ where for variables $a_i^{(k)}$, $b_j^{(l)}$ we have the standard Poisson structure:

$$\{a_i^{(k)}, a_j^{(l)}\} = \{b_i^{(k)}, b_j^{(l)}\} = 0, \ \{b_i^{(k)}, a_j^{(l)}\} = \delta_{kl}\delta_{ij}$$

Define the Hamiltonians

$$H_{\xi,*}^{(k)} = \sum_{i=1}^{N} a_i^{(k)}, b_i^{(k)} - \xi, \quad H_{\eta,i}^{(*)} = \sum_{k=1}^{n} a_i^{(k)}, b_i^{(k)} - \eta$$

A function on $(\mathcal{O}_{\xi_1} \times \cdots \times \mathcal{O}_{\xi_n})//A_N$ is called *polynomial* if it is the restriction of a polynomial A_N -invariant function on $(sl_N^*)^n$ to $\nu^{-1}(0) \subset \mathcal{O}_{\xi_1} \times \cdots \times \mathcal{O}_{\xi_n}$ where ν is the moment map for the diagonal Hamiltonian action of A_N .

It is easy to see that the space of polynomial functions on $(\mathcal{O}_{\xi_1} \times \cdots \times \mathcal{O}_{\xi_n})//A_N$ can be described as

(59)
$$Pol(\mathcal{O}_{\xi_1} \times \dots \times \mathcal{O}_{\xi_n} / / A_N) \simeq \langle a_i^{(k)}, b_j^{(l)} | \#(a_*^{(k)}) = \#(b_*^{(k)}), \\ \#(a_i^{(*)}) = \#(b_i^{(*)}), i = 1, \dots, N, k = 1, \dots, n > / \langle H_{\xi_k,*}^{(k)}, H_{\eta,i}^{(*)} \rangle,$$

where $\eta = (\xi_1 + \cdots + \xi_n)/N$. We can simplify the right side if we consider the action of the Cartan subgroup $H_N \times H_n \subset GL_N \times GL_n$ (here H_n are non-degenerate diagonal matrices) on the Poisson Heisenberg algebra generated by $a_i^{(k)}$ and $b_i^{(l)}$:

$$(h, h'): a_i^{(k)} \mapsto h_i h'_k a_i^{(k)}, \ b_j^{(l)} \mapsto h_j^{-1} h'_l^{-1} b_j^{(l)}$$

Then we can write (59) as

$$Pol(\mathcal{O}_{\xi_1} \times \cdots \times \mathcal{O}_{\xi_n} / / A_N) \simeq \langle a_i^{(k)}, b_j^{(l)} \rangle^{H_N \times H_n} / \langle H_{\xi_k, *}^{(k)}, H_{\eta, i}^{(*)} \rangle$$

The following theorem is an immediate consequence of this isomorphism.

THEOREM 5. We have the following isomorphism of symplectic varieties:

(60)
$$(\mathcal{O}_{\xi_1}^{SL_N} \times \cdots \times \mathcal{O}_{\xi_n}^{SL_N}) / / A_N \simeq (\mathcal{O}_{\eta}^{SL_n \times N}) (\xi_1, \dots, \xi_n) / / A_n$$

Here $\eta = (\xi_1 + \dots + \xi_n)/N$.

In terms of the variables $a^{(k)}$ and $b^{(k)}$ the Hamiltonian H_2 on $\mathcal{S}(\mathcal{O})_{reg}$ can be rewritten as

(61)
$$H_2 = \frac{1}{2} \sum_{i=1}^{N} p_i^2 - \sum_{i < j} \frac{\sum_{k,\ell=1}^{n} b_i^{(k)} a_j^{(k)} b_j^{(\ell)} a_i^{(\ell)}}{2 \mathrm{sh}^2 (q_i - q_j)}.$$

This Hamiltonian has a more natural "spin" interpretation. Now we have N particles and each of them have "internal degrees of freedom" or "spins" described by variables $\sum_{k=1}^{n} b_i^{(k)} a_j^{(k)}$ with $i \neq j$. In the next section we will give a better description of these variables in terms of coadjoint orbits.

3.3. Here we will rewrite the Hamiltonian (61) in terms of "spain" variables attached to one dimensional particles with the positions q_i , $i = 1, \ldots, N$. They are defined as follows.

Denote by g_{kl}^i the restriction of standard coordinate functions on sl_n^* to \mathcal{O}_η regraded as a function on k-th factor. We have:

$$g_{k\ell}^{(i)} = b_i^{(k)} a_i^{(\ell)} - \delta_{k\ell} \frac{\eta}{n}.$$

It is easy to check that if $i \neq j$ the following identity holds:

(62)
$$\sum_{k,\ell=1}^{n} g_{k\ell}^{(i)} g_{\ell k}^{(j)} = \mu_{ij} \mu_{ji} - \frac{\eta^2}{n}.$$

Thus, we can rewrite the Hamiltonian (58) in terms of spin variables from $\tilde{\nu}_{\tilde{\mathcal{O}}}^{-1}(t_{\xi})/\tilde{H} \times T^*A_{reg}$ as

(63)
$$H_2 = \frac{1}{2} \sum_{i=1}^{N} p_i^2 - \sum_{i < j} \frac{\operatorname{Tr}(g^{(i)}g^{(j)}) + \frac{\eta^2}{n}}{2\operatorname{sh}^2(\frac{q_i - q_j}{2})}.$$

This Hamiltonian describes N classical particles each carrying a "spin" variable which lies in the rank one coadjoint orbit in \mathfrak{sl}_n^* with the Casimir value given by

(64)
$$\sum_{k,l=1}^{n} g_{k,l}^{(i)} g_{l,k}^{(i)} = \eta^2 (1 - \frac{1}{n})$$

The system is Liouville integrable since we constructed n(N-1) integrals for the periodic spin chain earlier (see the proof of Theorem 2).

Integrable system described above are closely related [16] and [21].

3.4. This project, together with results of [1], is the first step towards constructing superintegrable systems on moduli spaces of flat connections on a surface where on part of the boundary the gauge group G is constrained to K. When the boundary gauge group is not constrained, corresponding integrable systems are described in [1]. We expect that such moduli spaces have the structure of a cluster variety similar to the one described in [14]. It would be interesting to extend the construction of spin CM chains to the elliptic case as it was done for N = 1 in [21].

Appendix A. Comparison with the n = 2 case from [31]

Consider the periodic spin CM chain from section 1 for n = 2. Let $G_2 = G \times G$ be the gauge group acting on $G \times G$ as it in section 1.1. Symplectic leaves of $T^*(G^{\times 2})/G_2$ are then

(65)
$$\mathcal{S}(\mathcal{O}_1, \mathcal{O}_2) = \{ (x_1, x_2, g_1, g_2) \mid x_1 - Ad_{g_2^{-1}}^*(x_2) \in \mathcal{O}_1, \\ x_2 - Ad_{g_1^{-1}}^*(x_1) \in \mathcal{O}_2 \} / G_2 \}$$

where $\mathcal{O}_1, \mathcal{O}_2$ are coadjoint orbits in \mathfrak{g}^* , relative to the gauge action

$$(h_1, h_2)(x_1, x_2, g_1, g_2) = (Ad_{h_1}^*(x_1), Ad_{h_2}^*(x_2), h_1g_1h_2^{-1}, h_2g_2h_1^{-1}).$$

In [31, §3 & App. C] the following Hamiltonian action of G_2 on $T^*(G^{\times 2})$ is considered,

(66)
$$(h_1, h_2)_*(x_1, x_2, g_1, g_2) = (Ad_{h_1}^*(x_1), Ad_{h_1}^*(x_2), h_1g_1h_2^{-1}, h_1g_2h_2^{-1}),$$

with corresponding moment map $\mu : T^*(C^{\times 2}) \to \mathfrak{g}^{*\times 2}$ given by

with corresponding moment map $\mu_*: T^*(G^{\times 2}) \to \mathfrak{g}^{*\times 2}$ given by

$$\mu_*(x_1, x_2, g_1, g_2) = (x_1 + x_2, -Ad_{g_1^{-1}}^*(x_1) - Ad_{g_2^{-1}}^*(x_2))$$

The corresponding symplectic leaves are

$$S_*(\mathcal{O}_1, \mathcal{O}_2) = \mu_*^{-1}(\mathcal{O}_1, \mathcal{O}_2)/G_2$$

= $\{(x_1, x_2, g_1, g_2) \mid x_1 + x_2 \in \mathcal{O}_1, -Ad_{g_1^{-1}}(x_1) - Ad_{g_2^{-1}}(x_2) \in \mathcal{O}_2\}/G_2,$

with the gauge group G_2 now acting by (66). These symplectic leaves were used in [31]. They are related to the symplectic leaves $\mathcal{S}(\mathcal{O}_1, \mathcal{O}_2)$ in the following way.

Consider the map $\psi: T^*(G^{\times 2}) \to T^*(G^{\times 2})$, defined by

$$\psi(x_1, x_2, g_1, g_2) = (-x_1, Ad_{g_1}^*(x_2), g_1, g_1g_2g_1).$$

Then ψ is G_2 -equivariant,

 $\psi((h_1, h_2)(x_1, x_2, g_1, g_2)) = (h_1, h_2)_* \psi(x_1, x_2, g_1, g_2),$

and the resulting map on the G_2 -orbits restricts to an isomorphism

$$\mathcal{S}(\mathcal{O}_1, \mathcal{O}_2) \xrightarrow{\sim} \mathcal{S}_*(\mathcal{O}_2, \mathcal{O}_1).$$

References

- S. Arthamonov, N. Reshetikhin, Superintegrable systems on moduli spaces of flat connections, Comm. Math. Phys. 386 (2021), 1337–1381. MR 4299124
- [2] G. Arutyunov, E. Olivucci, Hyperbolic spin Ruijsenaars-Schneider model from Poisson reduction, arXiv:1906.02619. MR 4133442
- [3] F. Calogero, Solution of the one-dimensional N-body problem with quadratic and/or inversely quadratic pair potentials, J. Math. Phys. 12 (1971), 419–436. MR 0280103
- [4] Calogero-Moser-Sutherland Models, J.-F. van Diejen, L. Vinet, Luc (Eds.), CRM Series in Mathematical Physics, Springer, 2000. MR 1843558
- [5] O. Chalykh, M. Fairon, On the Hamiltonian formulation of the trigonometric spin Ruijsenaars-Schneider system, Lett. Math. Phys. 110 (2020), 2893–2940. MR 4160924
- [6] Z. Chen, K. Jiang, H. Xiao and N. Reshetikhin, in preparation.
- [7] P. Etingof, O. Schiffmann, Twisted traces of intertwiners for Kac-Moody algebras and classical dynamical r-matrices corresponding to Belavin-Drinfeld triples, Math. Res. Lett. 6 (1999), 593-612. MR 1739218
- [8] P. Etingof, O. Schiffmann, *Lectures on quantum groups*. Lectures in Mathematical Physics. International Press, Somerville, MA, 2002. MR 2284964
- [9] P. Etingof, A. Varchenko, Traces of intertwiners for quantum groups and difference equations, I. Duke Math. J. 104 (2000), 391–432. MR 1781477
- [10] L. Feher, B.G. Pusztai, Spin Calogero models associated with Riemannian symmetric spaces of negative curvature, Nucl. Phys. B 751 (2006) 436–458. MR 2251936
- [11] L. Feher, B.G. Pusztai, A class of Calogero type reductions of free motion on a simple Lie group, Lett. Math. Phys. 79 (2007), 263–277. MR 2309780
- [12] L. Feher, An application of the reduction method to Sutherland type many-body systems, Geometric Methods in Physics, 109–117, Trends Math., Birkhauser/Springer, Basel, 2013. MR 3363996
- [13] G. Felder, Conformal field theory and integrable systems associated to elliptic curves, Proceedings of the International Congress of Mathematicians, Vol. 1,2 (Zurich, 1994), 1247–1255, Birkhauser, Basel, 1995. MR 1404026
- [14] V.V. Fock, A. Goncharov, Moduli spaces of local systems and higher Teichmüller theory, Publ. Math. Insti. Hautes Études Sci. No. 103 (2006), 1–211. MR 2233852
- [15] V.V. Fock, A.A. Rosly, Poisson structure on moduli of flat connections on Riemann surfaces and r-matrix, Moscow Seminar in Math. Phys., 67–86, Amer. Math. Soc. Transl. Ser. 2, 191, Adv. Math. Sci. 43, Amer. Math. Soc., Providence, RI, 1999. MR 1730456
- [16] J. Gibbons, T. Hermsen, A generalisation of the Calogero-Moser system, Phys. D 11 (1984), 337–348. MR 0761664
- [17] D. Kazhdan, B. Kostant, S. Sternberg, Hamiltonian group actions and dynamical systems of Calogero type, Comm. Pure Appl. Math. 31 (1978), 481–507. MR 0478225
- [18] S. Kharchev, A. Levin, M.A. Olshanetsky, A.A. Zotov, Calogero-Sutherland system with two types interacting spins, JETP Letters 106 (2017), 179–183.

- [19] S. Kharchev, A. Levin, M. Olshanetsky, A. Zotov, Quasi-compact Higgs bundles and Calogero-Sutherland systems with two types of spins, J. Math. Phys. 59 (2018), 103509, 36 pp. MR 3866479
- [20] A.W. Knapp, Lie Groups Beyond an Introduction, Progress in Math. 140, Birkhäuser, 1996. MR 1399083
- [21] I. Krichever, O. Babelon, E. Billey, M. Talon, Spin generalization of the Calogero-Moser system and the matrix KP equation. In: "Topics in topology and mathematical physics", 83–119, Amer. Math. Soc. Transl. Ser. 2, 170, Adv. Math. Sci., 27, Amer. Math. Soc., Providence, RI, 1995. MR 1355552
- [22] I.M. Krichever, A.V. Zabrodin, Spin generalization of the Ruijsenaars-Schneider model, the non-Abelian two-dimensional Toda lattice, and representations of the Sklyanin algebra, Uspekhi Mat. Nauk, 50 (1995), no. 6(306); Russian Math. Surveys 50 (1995), 1101–1150. MR 1379076
- [23] R. Sjamaar, E. Lerner, Stratified symplectic spaces, Annals of Mathematics 134, no. 2 (1991).
- [24] L.C. Li, P. Xu, Spin Calogero-Moser systems associated with simple Lie algebras, C.R. Acad. Sci. Paris Sér. I Math. 331 (2000), 55–60. MR 1780217
- [25] J. Moser, Three integrable Hamiltonian systems connected with isospectral deformations, Adv. Math. 16 (1975), 197–220. MR 0375869
- [26] N.N. Nekhoroshev, Action-angle variables, and their generalizations, Transl. Moscow Math. Soc. 26 (1972), 180–197. MR 0365629
- [27] N. Nekrasov, Holomorphic bundles and many-body systems, Comm. Math. Phys. 180 (1996), 587–603. MR 1408519
- [28] M.A. Olshanetsky and A.M. Perelomov, Classical integrable finite-dimensional systems related to Lie algebras, Phys. Rep. 71 (1981), 313–400. MR 0615898
- [29] N. Reshetikhin, Degenerate integrability of the spin Calogero-Moser systems and the duality with the spin Ruijsenaars systems, Lett. Math. Phys. 63 (2003), 55–71. MR 1967536
- [30] N. Reshetikhin, Degenerately integrable systems, J. Math. Sci. (N.Y.) 213 (2016), 769–785. MR 3493688
- [31] N. Reshetikhin, Spin Calogero-Moser models on symmetric spaces. In: "Integrability, quantization, and geometry. I. Integrable systems, 377–402, Proc. Sympos. Pure Math., 103.1, Amer. Math. Soc., Providence, RI, 2021. MR 4285685
- [32] N. Reshetikhin, Spin Calogero-Moser chains and two-dimensional Yang-Mills theory with corners., in preparation.
- [33] N. Reshetikhin, J.V. Stokman, Asymptotic boundary KZB operators and quantum Calogero-Moser spin chains. In: "Hypergeometry, integrability and Lie theory", 205–241, Contemp. Math., 780, Amer. Math. Soc., Providence, RI, 2022. MR 4476420
- [34] J.V. Stokman, Folded and contracted solutions of coupled classical dynamical Yang-Baxter and reflection equations, Indag. Math. (N.S.) 32 (2021), 1372–1411. MR 4334172
- [35] J.V. Stokman, N. Reshetikhin, N-point spherical functions and asymptotic boundary KZB equations, Invent. Math. 229 (2022), 1–86. MR 4438352
- [36] B. Sutherland, Exact results for a quantum many body problem in one dimension. II, Phys. Rev. A 5 (1972) 1372–1376.
- [37] M. Semenov-Tian-Shansky, Dressing transformations and Poisson group actions, Publ. Res. Inst. Math. Sci. 21 (1985), 1237–1260. MR 0842417
- [38] Superintegrability in Classical and Quantum Systems, Edited by: P. Tempesta, P. Winternitz, J. Harnad, W. Miller, Jr., G. Pogosyan, M. Rodriguez, CRM Proceedings and Lecture Notes 37 (2004). MR 2105429

YMSC, TSINGHUA UNIVERSITY, BEIJING, CHINA; BIMSA, BEIJING, CHINA; DEPARTMENT OF MATHEMATICS, UNIVERSITY OF CALIFORNIA, BERKELEY, USA; ST. PETERSBURG UNIVERSITY, RUSSIAN FEDERATION

Email address: reshetik@math.berkeley.edu